A Continuous-Time Modelling and Design Approach for Oversampled Passive Switched-Capacitor Filters

by

Sevil Zeynep Lüleç

A thesis submitted in conformity with the requirements for the degree of Doctor of Philosophy Graduate Department of Electrical & Computer Engineering University of Toronto

© Copyright 2019 by Sevil Zeynep Lüleç

Abstract

A Continuous-Time Modelling and Design Approach for Oversampled Passive Switched-Capacitor Filters

> Sevil Zeynep Lüleç Doctor of Philosophy Graduate Department of Electrical & Computer Engineering University of Toronto

> > 2019

Integrated technology advances towards shorter length transistors with faster operation frequencies and lower supply and threshold voltages. While these trends are pushed by digital performance, analog and RF designs do not necessarily benefit from them. Traditional high gain and high bandwidth operational amplifiers become harder to design, and thus topologies based on "digital friendly" components such as switches and capacitors can become preferable. In terms of filter design, these trends favour passive switchedcapacitor networks, where active components (such as operational amplifiers) are not needed to transfer charge in between capacitors.

This thesis proposes a simplified continuous-time approach for modelling, design, analysis, and simulation of oversampled switched-capacitor circuits focusing on the passive switched-capacitor filters. The model provides an intuitive understanding of these structures by letting designers reach transfer functions and noise characteristics with good accuracy and without the need to go through complicated charge-balance equations. The aim of the model is mainly to give designers a tool that can lead to innovations through intuitive understanding. It is applicable to a variety of topologies including multi-phase passive switched-capacitor filters, switched-capacitor integrators, as well as switched-capacitor DC/DC converters.

With the proposed model, for the first time, complex-conjugate poles are integrated

on silicon by using only switches and capacitors. A continuous-time approach is reported to implement low-pass transfer functions with a sharper frequency profile compared to the passive switched-capacitor topologies present in the literature. Theory and simulation results are verified with measurements performed on the filter prototype integrated in a 0.13 µm CMOS technology. The prototype has a cut-off frequency of 470 kHz, 150 µW power consumption from a 1.2 V power supply, 92 dB SFDR, and an active area of $0.06 \ mm^2$.

to Andaç

Contents

1	Bac	kgrour	nd	1
	1.1	A Very	y Brief History	1
		1.1.1	Switched-Capacitor Circuits	1
		1.1.2	Passive Switched-Capacitor Circuits	3
	1.2	Renew	red Interest	4
		1.2.1	Integration Sampler	5
		1.2.2	2^{nd} -Order Discrete-Time IIR Low-Pass Filter	7
		1.2.3	High-Order Discrete-Time IIR Low-Pass Filter	8
		1.2.4	Passive Switched-Capacitor Filter with Active Feedback \ldots .	9
	1.3	Contin	uous-Time Modelling	10
	1.4	Motiva	ation	12
		1.4.1	Organization of the Thesis	13
2	Froi	n Disc	rete to Continuous-Time Modelling	14
	2.1	Propos	sed Continuous-Time Model	15
		2.1.1	Theory	15
		2.1.2	Reciprocity	15
		2.1.3	Examples	21
	2.2	The C	ontinuous-Time Model for Non-Grounded Sampling Capacitors	24
		2.2.1	Theory	26

		2.2.2	Examples	27
3	Cor	nplex-0	Conjugate Poles using Switches and Capacitors	31
	3.1	Passiv	e Switched-Capacitor Butterworth Biquad	31
		3.1.1	From Real Poles to Complex-Conjugate Poles	31
		3.1.2	Implementation of the Passive Switched-Capacitor Filter	33
		3.1.3	Filter Transfer Function Simulations	37
		3.1.4	Effect of Parasitic Capacitances	38
		3.1.5	Single Pole Passive Switched-Capacitor Complex Filter	38
	3.2	High-(Order Filtering with Complex-Conjugate Poles	41
		3.2.1	Continuous-Time Design and Optimization	41
		3.2.2	Implementation of the High-Order Passive Switched-Capacitor Filter	42
4	Noi	se Ana	lysis and Practical Limitations of the Model	46
	4.1	Settlin	ng Time	46
	4.2	Charge	e Sharing	47
	4.3	Accura	acy of the Model in PSC Filter Response	52
	4.4	Noise	Analysis and Simulation Using the Model	52
		4.4.1	Examples	53
5	Pro	totype	Filter Design and Implementation	61
	5.1	Protot	ype Filter Differential Implementation	61
		5.1.1	Component Mismatch Effect on the Quality Factor	64
	5.2	Measu	rement Results	64
6	Cor	nclusio	ns	71
	6.1	Contri	butions	72
	6.2	On-Go	ping and Future Work	73
		6.2.1	Increasing Configurability and Linearity of the PSC Filter	73

	6.2.2	Band-Pass and High-Pass Passive Switched-Capacitor Filters	74
	6.2.3	Passive Switched-Capacitor Digital-to-Analog Converter	75
	6.2.4	High-Order Butterworth Passive Switched-Capacitor Filter	75
A	Equivalen	t Resistance Derivation for Low Frequency	77
в	Continuou	s-Time Model Derivation for Multi-Phase Topology	81
С	The Conti	nuous-Time Model with Capacitance	84
Bi	Bibliography		

List of Figures

1.1	A condenser whose connections are reversed at regular intervals of time [1].	2
1.2	1^{st} -order PSC filter with its clocking scheme [2]	3
1.3	3^{rd} -order IIR Butterworth filter magnitude response compared with a)	
	125-tap FIR filter and b) 260-tap FIR filter.	6
1.4	a) 1^{st} -order anti-aliasing integration sampler, and b) its transfer function [2].	7
1.5	2^{nd} -order PSC filter with three clock phases [3, p. 228]	7
1.6	N^{th} -order PSC filter with all real poles [4]	9
1.7	N^{th} -order PSC filter with active feedback. [5]	10
1.8	Basic two-phase switched-capacitor topology modelled with the traditional	
	equivalent resistance [6, p. 399]	11
1.9	a) 1^{st} -order PSC filter and b) its continuous-time model with the tradi-	
	tional equivalent resistance approach $[2]$	11
2.1	Basic two-phase switched-capacitor topology modelled with the traditional	
	equivalent resistance [6, p. 399]	14
2.2	A reciprocity test for a continuous-time two-port network [7, p. 4]	15
2.3	A reciprocity test for the two-phase switched-capacitor circuit	16
2.4	A reciprocity test for a three-phase switched-capacitor circuit	17
2.5	Two-phase switched-capacitor topology a) with the average current into	
	V_2 is modelled, b) with the average current into V_1 is modelled, and c)	
	with its complete proposed continuous-time model	20

2.6	a) 2^{nd} -order PSC filter with three clock phases [3, p. 228], b) its continuous-	
	time model	22
2.7	$2^{nd}\mbox{-}{\rm order}$ PSC filter simulated discrete-time implementation and continuous-	
	time model frequency responses with logarithmic and linear scale frequency	
	axes	22
2.8	$2^{nd}\mbox{-}{\rm order}$ PSC filter simulated discrete-time implementation and continuous-	
	time model step responses, and the error between the step responses. $\ .$.	23
2.9	a) N^{th} -order PSC filter with $N+1$ clock phases [4], b) its continuous-time	
	model	24
2.10	a) Parasitic insensitive switched-capacitor integrator, and b) its incorrect	
	continuous-time model	25
2.11	a) Two-phase switched-capacitor topology with non-grounded sampling	
	capacitor and b) its proposed continuous-time model	26
2.12	Parasitic insensitive switched-capacitor integrator a) switching parts, b)	
	its continuous-time model, and c) rearranged continuous-time model	27
2.13	The continuous-time model of the parasitic insensitive switched-capacitor	
	integrator	28
2.14	a) Switched-capacitor DC/DC voltage converter, b) its simplified continuous-	
	time model, c) the Thévenin equivalent of v_{out} obtained with the continuous-	
	time model, d) conventionally used switched-capacitor converter trans-	
	former model.	30
31	a) Passive switched-capacitor filter with two real poles and b) its continuous-	
0.1	time model	32
२	Low-pass filter with complex-conjugate poles involving a negative feedback	02
0.2	loop	33
33	Differential 2^{nd} -order passive switched-capacitor filter with complex conjugate	55
J.J	polos	21
	hotee	94

3.4	Differential 2^{nd} -order Butterworth biquad continuous-time model: a) Di-	
	rect representation of four phases, b) with parallel branches merged, c)	
	cross-coupled buffers are replaced by inverter branches	35
3.5	Simulated magnitude responses for $1dB$ droop at $1MHz$ with $f_S = 160MHz$	
	of the two-real-pole and 2^{nd} -order Butterworth biquads	37
3.6	Single pole complex PSC filter a) discrete-time implementation [8] and b)	
	its continuous-time model	40
3.7	Simulated normalized frequency responses of PSC complex filter discrete-	
	time implementation and its continuous-time model	40
3.8	The proposed high-order low-pass PSC filter with complex-conjugate poles.	42
3.9	Pole locations of the 2^{nd} , 3^{rd} , and 4^{th} -order filters, where the gray circles	
	mark the same order Butterworth pole locations having the same $1dB$	
	cut-off frequency.	43
3.10	The continuous-time model of the 3^{rd} -order filter with complex-conjugate	
	poles	44
3.11	The PSC implementation of the 3^{rd} -order filter with complex-conjugate	
	poles	44
3.12	The transfer function of the PSC filter with complex-conjugate poles com-	
	pared with the same order Butterworth and the 7^{th} -order real pole filter	
	having the same $1dB$ droop frequency.	45
4.1	Basic two-phase switched-capacitor topology with series resistances	47
4.2	a) 1^{st} -order PSC filter and b) its continuous-time model with the tradi-	
	tional equivalent resistance approach [2]	48
4.3	1^{st} -order PSC filter discrete-time implementation and continuous-time model	
	simulated frequency responses for $C_I/C_S = 40$ and $C_I/C_S = 4$ with loga-	
	rithmic and linear scale frequency axes	49

4.4	1^{st} -order PSC filter discrete-time implementation and continuous-time model	
	simulated step responses for $C_I/C_S = 40$ and $C_I/C_S = 4$	50
4.5	Error in the $3dB$ cut-off frequency calculation of a 1^{st} -order PSC filter	
	using the continuous-time model for changing a) C_S/C_I and b) f_{3dB}/f_S	
	ratios	51
4.6	Correlated noise charge stored in two capacitors with opposite polarities.	53
4.7	2^{nd} -order PSC filter and its continuous-time model	53
4.8	2^{nd} -order PSC filter and its continuous-time model with noise sources,	
	where the two ϕ_1 noise sources are fully correlated with inverse polarity	54
4.9	$2^{nd}\text{-}\mathrm{order}$ PSC filter discrete-time noise analyzes signal flow diagram. 	55
4.10	The continuous-time model of the 2^{nd} -order PSC filter for noise simulation.	57
4.11	Simulated output noise power spectral densities of 2^{nd} -order PSC filter	
	and its continuous-time model	57
4.12	The 2^{nd} -order complex-conjugate-pole filter proposed model with noise	
	sources, where noise sources associated with the same switch are fully	
	correlated with inverse polarity	58
4.13	Simulated output noise spectral densities of the two-real-pole and 2^{nd} -order	
	Butterworth biquads for $1dB$ droop at $1MHz$ with $f_S = 160MHz$	59
4.14	2^{nd} -order PSC filter and its continuous-time model with noise sources,	
	where two ϕ_1 noise sources are fully correlated with an inverse polarity	60
5.1	The fabricated 3^{rd} -order PSC low-pass filter schematic and the timing	
	diagram.	62
5.2	The fabricated 3 rd -order PSC low-pass filter chip micrograph.	65
5.3	Transfer functions of the simulated and measured 3^{rd} -order low-pass PSC	
0.0	filter and the simulated 7^{th} -order all real pole filter.	66
5.4	Measured low-pass prototype filter transfer functions for changing sam-	
	pling frequencies. $f_{\rm s}$.	66
	r o 1	

5.5	Measured low-pass prototype filter transfer functions for 1^{st} , 2^{nd} , and 3^{rd}	
	-order output nodes for $f_S = 160 MHz$	67
5.6	Measured output noise spectral density over the prototype filter bandwidth.	68
5.7	Out-of-band IIP3 measurements of the prototype filter with two tones. $% \mathcal{A}^{(1)}$.	68
6.1	a) A symbolic passive switched-capacitor filter with a gain of 2 charge	
	pump, and b) its continuous-time model	73
6.2	A voltage input voltage output passive switched-capacitor band-pass filter	
	continuous-time model and its discrete-time implementation	74
6.3	The basic idea for a passive switched-capacitor DAC with 3^{rd} -order recon-	
	struction filtering.	76
6.4	Passive switched-capacitor filter continuous-time model with multiple feed-	
	back loops	76
B.1	A three-phase switched-capacitor topology, and b) its continuous-time	
	model	82
C.1	a) 1^{st} -order PSC filter, b) its equivalent resistance continuous-time model,	
	and c) its equivalent resistance continuous-time model with $C_{\rm S}/2$ capaci-	
	tance added	85
C.2	1^{st} -order PSC filter simulated frequency responses of discrete-time imple-	
	mentation and continuous-time models with and without $C_S/2$ for $C_I/C_S =$	
	40 and $C_I/C_S = 4$ (logarithmic scale frequency axis on top and linear scale	
	frequency axis on bottom)	85
C.3	1^{st} -order PSC filter simulated step responses of discrete-time implementa-	
	tion and continuous-time models with and without $C_S/2$ for $C_I/C_S = 40$	
	and $C_I/C_S = 4$	87

Chapter 1

Background

1.1 A Very Brief History

1.1.1 Switched-Capacitor Circuits

In the literature, the switched-capacitor implementation and its behaviour as an equivalent resistance can be found as early as 1873, in the book of "A Treatise on Electricity and Magnetism" by J. C. Maxwell [1]. Maxwell states that if the connections of a condenser, which is connected in between the ends of a wire, are reversed at regular intervals of time, T, then the quantity of electricity transmitted by the wire in each interval is equal to 2EC, where E is the electromotive force, and C is the capacity of the condenser. Fig. 1.1 visualizes this behaviour. He also states that this behaviour is only valid if the interval T is sufficiently long to allow of the complete discharge of the condenser. Thus, the condenser with the periodically reversing connections acts like an equivalent resistor:

$$\frac{E}{R} = \frac{2EC}{T}; \quad \text{or} \quad R = \frac{T}{2C}.$$
(1.1)

Approximately 100 years after Maxwell's book, switched-capacitor implementations reached popularity in electronics design due to the invention of MOS integrated cir-



Figure 1.1: A condenser whose connections are reversed at regular intervals of time [1].

cuits [9–16]. Traditionally used active-RC topologies suffered from inaccurate transfer functions due to the mismatch in between integrated resistor and capacitor values (such as filter corner frequencies can vary up to 40 % over process and temperature variations [17]). Passive filters, which are composed of capacitors and inductors, were not an alternative as inductors were not possible to integrate at that time. On the other hand, switched-capacitor based filters became attractive due to their characteristics depending on capacitor ratios, which have higher accuracy for fabricated devices (down to 0.1 % [18]). Fabricated devices based on MOS technology decreased required cost and area of the circuits, and thus switched-capacitor based circuits and more specifically filters gained popularity. Since then, switched-capacitor filters have been widely used in signal processing by replacing the resistors in the equivalent active-RC implementations [6, p. 417]. Moreover, switched-capacitor based circuits have been used in various integrated circuits such as analog-to-digital (A/D) [19, 20] and digital-to-analog (D/A) converters [21, 22], mixers and samplers [23, 24], and power converters [25, 26].

Issues and the Return of Continuous-Time Filters

Inevitably, switched-capacitor circuits came along with some limitations. In terms of design, the synthesis of filters in z-domain was challenging for circuit designers. Moreover, several issues accompanying switched-capacitor circuits such as signal and noise aliasing, charge injection, opamp settling time, and slew rate limitations led designers to continue to work on improving the accuracy of continuous-time filters. There were many developments in the area of tunable and programmable continuous-time filters, which



Figure 1.2: 1^{st} -order PSC filter with its clocking scheme [2].

include active-RC [27], MOSFET-C [28], and gm-C [29,30] filters that can address some of the issues of the early continuous-time filter attempts. Afterwards, around the 1990s, inductors became practical at 1GHz and above, leading to the use of passive filters.

Today, both continuous-time filters and switched-capacitor filters find applications in integrated circuit design [31–35], [36–40]. Switched-capacitor filters are preferred due to their accurate transfer functions depend on capacitor ratios and clock frequency with no requirement for calibration and tuning over changing process-voltage-temperature (PVT) variations. However, the existing operational amplifier for the charge transfer and requirement of oversampling can limit the speed of these filters.

1.1.2 Passive Switched-Capacitor Circuits

Passive switched-capacitor (PSC) circuits, whose variations can be found early as 1978 [41–48], are a particular subset of switched-capacitor networks that do not rely on active elements to transfer charge between capacitors [3]. As a basic example, the current input voltage output 1st-order filter is shown in Fig. 1.2 can be considered as a passive switched-capacitor structure as there is no active component. In other words, there is no amplifier used to transfer charge between capacitors C_I and C_S . (Although, the switch is generally implemented using a transistor, the transistor is biased in triode (linear) region. Thus, it acts as a passive element, i.e., resistor.)

Elimination of the active elements may allow one to increase the maximum operating frequency up to the settling time required by the passive charge sharing paths, to reduce

CHAPTER 1. BACKGROUND

the consumed power, and to improve the linearity due to the fewer number of active components. PSC filters also inherit the advantages of switched-capacitor filters such as the corner frequencies are defined as capacitors ratios. This property also leads to PVT independent corner frequencies. Moreover, similar to switched-capacitor filters, corner frequencies can also be easily programmable by changing the clock frequency [3].

In the 1980s, in parallel to the advances in switched-capacitor topologies, passive switched-capacitor structures also had their fair share of interest. The advantages of the passive switched-capacitor filters were mainly pointed out as the lower power supply and power consumption requirements due to the elimination of the operational amplifiers. Moreover, the simple structure of filters that are only composed of switches and capacitors were easy to implement using integrated technology. During that time, various filter and even resonator structures with complex-conjugate poles were studied [41, 44]. However, having no operational amplifiers resulted in limitations on the number of cascaded stages as well as limiting the quality factor (Q) of the poles. Thus, realizable filter characteristics become restricted [44].

1.2 Renewed Interest

As integrated circuit technology advances towards shorter length transistors, analog designers need to reconsider the conventional implementations. Advanced transistors require lower power supply voltages and make the design of high-gain high-bandwidth amplifiers challenging to design. However, switch and capacitor based circuits work well in modern technologies due to lower parasitic components and fast clocking circuits. In terms of filter design, these trends favour passive switched-capacitor architectures, where the charge transfer between capacitors do not require the presence of an active element [3]. Passive switched-capacitor circuits are discrete-time circuits that are closest to the digital implementation. They are composed of switches, capacitors, and phase generators, all of

CHAPTER 1. BACKGROUND

which benefit technology scaling. They are generally driven by inverter-based transconductors (g_m -cells), which are compatible with digital technology with improving g_m over bias current [4,5,49]. Moreover, these passive switched-capacitor infinite impulse response (IIR) filters can be advantageous compared to the finite impulse response (FIR) implementations, especially for large oversampling ratio applications. For instance, consider an oversampled filter with a sampling frequency of 160 MHz and 3 dB cut-off frequency of 500 kHz. An example FIR filter design can need more than 100 taps to approximate the transfer function of a $3^r d$ -order IIR Butterworth filter (Fig. 1.3a). Although this design achieves around 40 dB attenuation and mimics the Butterworth behaviour close to the band-edge, it fails to achieve high attenuation values at higher frequencies. If FIR filter design is modified to match also the high attenuations at higher frequencies, then we can end up with a more than 250-tap filter (Fig. 1.3b).

In recent literature, passive switched-capacitor topologies were used as an anti-aliasing integration sampler [2, 24], a channel-selection filter for receivers [4, 5, 8, 50–54], a decimation filter before an ADC [55], and a charge-based DAC for transmitters [21]. This section covers the operation and properties of passive switched-capacitor filters from recent literature.

1.2.1 Integration Sampler

The integration sampler shown in Fig. 1.4a was employed in a software-defined wireless receiver prototype to create discrete samples of the down-converted signal before fed into an ADC [24]. This 1st-order sampler integrates the input signal on an integration capacitor, C_I , for a sampling period of T_S before sampling it on C_S with a gain of $g_m/(C_I + C_S)$. The integration sampling results in notches at the integer multiples of the sampling frequency, f_S (Fig. 1.4b). These notches attenuate the blockers that are folding on the desired channel. Moreover, C_I and C_S form a discrete-time infinite impulse response (IIR) filter with a single real pole at $z_p = C_I/(C_I + C_S)$. This pole leads to the



Figure 1.3: 3^{rd} -order IIR Butterworth filter magnitude response compared with a) 125-tap FIR filter and b) 260-tap FIR filter.



Figure 1.4: a) 1^{st} -order anti-aliasing integration sampler, and b) its transfer function [2].



Figure 1.5: 2^{nd} -order PSC filter with three clock phases [3, p. 228].

attenuation of the adjacent channels due to the discrete-time IIR low-pass filter behaviour (Fig. 1.4b). Attenuation of the side-lobes releases the dynamic range requirements of the following ADC. Since both anti-aliasing notches and discrete pole depends on f_S , this integration sampler can be configurable to handle different bands only by changing f_S [2]. This sampler shows that the continuous input current integration on the filter capacitor, C_I , prior to sampling results in build-in 1st-order anti-aliasing filtering, which is a property that is also inherited by subsequent high-order filters.

1.2.2 2nd-Order Discrete-Time IIR Low-Pass Filter

The anti-aliasing integration sampler discussed above can only realize 1^{st} -order filtering. However, it is possible to modify the integration sampler to have a 2^{nd} -order discrete-time IIR low-pass filter characteristics by connecting C_S to a second integrating capacitor, C_{I2} , before resetting it to ground (Fig. 1.5) [3, p. 228]. The transfer function of the filter can be calculated first by writing charged-domain equations as follows:

$$(C_{I1} + C_S)V_1[n] = C_{I1}V_1[n-1] + Q_{in}[n]$$

$$(C_{I2} + C_S)V_{out}[n+1/3] = C_{I2}V_{out}[n-2/3] + C_SV_1[n+1/3]$$
where $Q_{in}[n] = \int_{(n-1)T_S}^{nT_S} g_m V_{in}(t)dt$
(1.2)

where n represents any sampling period with ϕ_1 , ϕ_2 , and ϕ_3 , and n-1 represents the preceding sampling period. From Eqn. 1.2, z-domain equations can be written as:

$$(C_{I1} + C_S)V_1(z) = C_{I1}V_1(z)z^{-1} + Q_{in}(z)$$

$$(C_{I2} + C_S)V_{out}(z)z^{1/3} = C_{I2}V_{out}(z)z^{-1}z^{1/3} + C_SV_1(z)z^{1/3}$$
(1.3)

Then, the z-domain transfer function becomes:

$$\frac{V_{out}(z)}{Q_{in}(z)} = \frac{1}{C_S} \frac{(1-\alpha_1)(1-\alpha_2)}{(1-\alpha_1 z^{-1})(1-\alpha_2 z^{-1})}$$
(1.4)

where $\alpha_i = C_{Ii}/(C_{Ii} + C_S)$. From the transfer function, it can be seen that there exist two real poles at $z_{p1} = \alpha_1$ and $z_{p1} = \alpha_2$. The DC gain can also be found as $1/C_S$ for z = 1. The bilinear approximation can be used to find the corresponding s-domain pole locations as:

$$s_{pi} = \frac{2}{T_S} \frac{z_{pi} - 1}{z_{pi} + 1} = \frac{2}{T_S} \frac{\alpha_i - 1}{\alpha_i + 1} = \frac{2}{T_S} \frac{-C_S}{2C_{Ii} + C_S} \bigg|_{C_{Ii} \gg C_S} \approx -\frac{C_S}{T_S C_{Ii}}$$
(1.5)

1.2.3 High-Order Discrete-Time IIR Low-Pass Filter

To further increase the order of the low-pass passive switched-capacitor filter described in the previous part, a passive charge rotating filter structure was reported [4,50]. It is possible to increase the order of the filter by adding more integrating capacitors and clock phases, such that a single C_S shares its charge with integrating capacitors sequentially before being reset to the ground in the last phase (Fig. 1.6). The number of the inte-



Figure 1.6: N^{th} -order PSC filter with all real poles [4].

grating capacitors in the filter determines the number of real poles created. The transfer function can be written similar to the 2^{nd} -order low-pass filter, starting from the charge balance equations (1.2). Then, the z-domain transfer function becomes:

$$\frac{V_{out}(z)}{Q_{in}(z)} = \frac{1}{C_S} \frac{(1-\alpha_1)(1-\alpha_2)\dots(1-\alpha_N)}{(1-\alpha_1 z^{-1})(1-\alpha_2 z^{-1})\dots(1-\alpha_N z^{-1})}$$
(1.6)

where $\alpha_i = C_{Ii}/(C_{Ii} + C_S)$. From the transfer function, it can be seen that there exist N real poles at $z_{pi} = \alpha_i$ with $i \in \mathbb{Z}[1, N]$. The DC gain can also be found as $1/C_S$ for z = 1. The bilinear approximation can be used to find the corresponding s-domain pole locations as:

$$s_{pi} = \frac{2}{T_S} \frac{z_{pi} - 1}{z_{pi} + 1} = \frac{2}{T_S} \frac{\alpha_i - 1}{\alpha_i + 1} = \frac{2}{T_S} \frac{-C_S}{2C_{Ii} + C_S} \bigg|_{C_{Ii} \gg C_S} \approx -\frac{C_S}{T_S C_{Ii}}$$
(1.7)

This high-order filter can only create real poles.

1.2.4 Passive Switched-Capacitor Filter with Active Feedback

Complex-conjugate pole implementation leads to sharper roll-off around band-edge and can lead to lower filter order for the same band-pass and band-stop specifications. A recent work adds an active feedback path from the output node to the input node of the passive switched-capacitor filter discussed above to generate complex-conjugate poles



Figure 1.7: N^{th} -order PSC filter with active feedback. [5]

(Fig. 1.7) [5]. Although this approach achieves better roll-off close to the band edge due to the realized complex-conjugate poles, it eliminates some of the fundamental properties of the passive switched-capacitor filters. Most importantly, the transconductance realizes the active feedback leads to filter characteristics that are no longer only depends on capacitor ratios. Moreover, the feedback transconductance adds additional noise and non-linearity to the system.

1.3 Continuous-Time Modelling

Switched-capacitor circuits have linear time-varying characteristics requiring charge balance equations and z-domain analysis for the exact representation of their behaviour in discrete-time [6, p. 398]. However, under specific conditions, it is possible to analyze these structures with good accuracy using linear time-invariant components. For instance, as mentioned earlier, it is well known that a capacitor, C_S , connected to two voltage sources V_1 and V_2 in two different clock phases, ϕ_1 and ϕ_2 , respectively, can be modelled using an equivalent resistance as shown in Fig. 1.8 (where R is equal to T_S/C_S with T_S corresponds to the clock period) [6, p. 399] [1]. This equivalence is derived by assuming ideal DC voltage sources connected to V_1 and V_2 nodes and by calculating the average current transfer in one clock period.



Figure 1.8: Basic two-phase switched-capacitor topology modelled with the traditional equivalent resistance [6, p. 399].



Figure 1.9: a) 1^{st} -order PSC filter and b) its continuous-time model with the traditional equivalent resistance approach [2].

This simple modelling approach helps one intuitively understand the operation of oversampled switched-capacitor circuits. For example, Fig. 1.9a shows the integration sampler (shown earlier in Fig. 1.4a) with current input and voltage output by focusing on the 1st-order filtering. In this filter, C_S is switching in between a grounded integrating capacitor, C_I , and the ground node. C_I can be treated as a voltage source for C_I values much larger than C_S . Thus, the switching parts can be replaced by an equivalent resistor with a value of $1/f_S C_S$ (Fig. 1.9b) [2]. By using this model, the approximate DC gain ($R = 1/f_S C_S$) and bandwidth of the filter ($\omega_{3dB} = 1/RC_I = f_S C_S/C_I$) can be written intuitively without the need of charge-balance equations and z-domain transfer functions. However, there is no intuitive way to apply the equivalent resistor approach to the three-phase 2nd-order switched-capacitor filter, which was shown in Fig. 1.5.

1.4 Motivation

Complex-Conjugate Poles

The passive switched-capacitor filters mentioned in the previous section rely on real poles preventing the synthesis of sharp filtering profiles such as the ones obtained by using Butterworth and Chebyshev filters. The one filter that can produce sharp filtering profile uses active feedback, which eliminated the advantages of the passive structure. In the literature, there also exists a complex passive switched-capacitor filter; however, it can only create a single complex pole by making use of an in-phase and a quadrature current inputs with two g_m -cells and four phases [8]. Implementation of complex-conjugate poles to realize Butterworth and Chebyshev filter responses may result in better suppression of the blockers compared to the real pole filters with the same orders. These sharper filtering profiles may help, for example, releasing the dynamic range requirements of the following stages. Moreover, it can result in lower power consumption as the filter order can be decreased for the same selectivity of the filter.

A Simplified Continuous-Time Model

Another issue with passive switched-capacitor filters covered so far is that the analysis of such filters is not intuitive and requires tedious charge-balance equations to obtain z-domain transfer function. Although rigorous analyses do not create a problem for the design of known structures, since it is a one-time effort, it can complicate the invention of new topologies. Moreover, noise analysis becomes even more complicated and requires long simulation times. A simplified continuous-time model that addresses these issues can lead to an intuitive understanding of the filter operation as well as the invention of new structures.

1.4.1 Organization of the Thesis

This thesis is organized as follows: Chapter 2 describes the proposed continuous-time modelling approach that can address multi-phase oversampled switched-capacitor structures together with examples. Chapter 3 shows how to implement complex-conjugate poles using the proposed model and only with switches and capacitors. Chapter 4 describes the noise analysis using the proposed model and goes through some of the practical limitations. Chapter 5 covers the implementation of the 3^{rd} -order passive switched-capacitor filter prototype with complex-conjugate poles. Finally, Chapter 6 summarizes contributions, on-going, and future work.

Chapter 2

From Discrete to Continuous-Time Modelling

In this section, the proposed continuous-time modelling of the oversampled switchedcapacitor circuits is explained. A new model is developed, starting from the traditional equivalent resistance, for non-reciprocal multi-phase switched-capacitor structures, leading to an intuitive modelling and design approach. Several examples are also included to demonstrate the modelling approach and to address common misconceptions.



Figure 2.1: Basic two-phase switched-capacitor topology modelled with the traditional equivalent resistance [6, p. 399].



for reciprocity $i_{t1} = i_{t2}$

Figure 2.2: A reciprocity test for a continuous-time two-port network [7, p. 4].

2.1 Proposed Continuous-Time Model

2.1.1 Theory

It is known that a capacitor, C_S , connected to two voltage sources V_1 and V_2 in two different clock phases, ϕ_1 and ϕ_2 , respectively, can be modelled using an equivalent resistance, R, as shown in Fig. 2.1 and as discussed in Section 1.1. Although this model can lead to an intuitive understanding of oversampled switched-capacitor structures [6, p. 399], it is limited to two-phase topologies with reciprocal low frequency behaviour.

2.1.2 Reciprocity

The reciprocity theorem is traditionally defined for continuous-time circuits. However, the reciprocity definition can come in handy to understand the shortcomings and applicability of the switched-capacitor equivalent resistor model. A continuous-time two-port network is defined as reciprocal if a test voltage source, v_t , applied to the first port generates a short circuit current on the second branch, i_{t1} , which is the same as the short circuit current generated on the first branch, i_{t2} , if the same test voltage source, v_t , is applied to the second port, as shown in Fig. 2.2 (i.e., $Z_{12} = Z_{21}$). A network that contains only ideal R, L, and C elements is reciprocal [7, p. 6]. In light of the reciprocity definition, we can conclude that a network of equivalent resistors can only model switched-capacitor topologies with a reciprocal low frequency behaviour.

Switched-capacitor circuits are time-varying circuits. Thus, a reciprocity test applied



Figure 2.3: A reciprocity test for the two-phase switched-capacitor circuit.

on a switched-capacitor circuit fails. However, we are interested in the low frequency behaviour of these structures. We want to understand if this behaviour is "reciprocal" or not, such that we can have more clues on how to model these structures correctly.

Fig. 2.3 shows one "modified" reciprocity test applied to the conventional two-phase switched-capacitor circuit (Fig. 2.1). In this "modified" test, applied voltages are DC sources, and observed currents are average currents. Thus, effectively, we are checking the reciprocity of these circuits for DC signals.

Two average currents produced $(i_{t1} \text{ and } i_{t2})$ as the result of the test voltage (v_t) applied to nodes V_1 and V_2 , respectively are derived to be equal to each other:

$$i_{t1} = \frac{\Delta Q}{T_S} = \frac{C_S V_t}{T_S} \tag{2.1}$$

$$i_{t2} = \frac{\Delta Q}{T_S} = \frac{C_S \cdot V_t}{T_S} \tag{2.2}$$

where T_S is the sampling period. Thus, the two-phase switched-capacitor circuit can be defined to have a reciprocal low frequency behaviour. (It should be noted here that although the averaged behaviour is checked for simplicity, it is also possible to extend this approach for low frequency behaviour, where the frequency of interest is much smaller than the sampling frequency.) Thus, using an equivalent resistor to model the reciprocal low frequency behaviour of a two-phase switched-capacitor circuit makes sense.

Now, we can check the reciprocity of the three-phase switched-capacitor circuit shown



Figure 2.4: A reciprocity test for a three-phase switched-capacitor circuit.

in Fig. 1.5. Fig. 2.4 shows the reciprocity test applied to the switching parts of the circuit. Two average currents produced $(i_{t1} \text{ and } i_{t2})$ as the result of test voltage (v_t) applied to nodes V_1 and V_{out} , respectively can be derived as follows:

$$i_{t1} = \frac{\Delta Q}{T_S} = \frac{C_S \cdot V_t}{T_S} \tag{2.3}$$

$$i_{t2} = \frac{\Delta Q}{T_S} = \frac{C_S . 0}{T_S} = 0 \tag{2.4}$$

where T_S is the sampling period. Although i_{t1} has the same non-zero value as Eqn. 2.2, i_{t2} has a zero value since C_S is reset to the ground in ϕ_3 . Thus, the three-phase switchedcapacitor circuit can be defined to have a non-reciprocal low frequency behaviour. Thus, we can conclude that using only bilateral elements such as R, L, and C to model the low frequency behaviour of this multi-phase circuit do not work out.

The Proposed Unilateral Model

To develop a new continuous-time model that can address multi-phase topologies with non-reciprocal low frequency behaviour, we can start by examining the derivation of the traditional equivalent resistance model in detail.

The resistance equivalence is derived by assuming ideal DC voltage sources connected to V_1 and V_2 nodes and by calculating the average current transfer in one clock period as follows:

$$\Delta Q = C_S (V_1 - V_2) \tag{2.5}$$

$$I_A = \frac{\Delta Q}{T_S} = \frac{C_S (V_1 - V_2)}{T_S}$$
(2.6)

$$R = \frac{V_1 - V_2}{I_A} = \frac{T_S}{C_S} = \frac{1}{f_S \cdot C_S}$$
(2.7)

where I_A corresponds to the average current flowing from V_1 node to V_2 node in one sampling period, T_S . It is straight forward to show that the average current flowing from node V_1 is equal to the average current flowing into node V_2 . Thus, a single $1/f_S C_S$ valued resistance is enough to model the average current behaviour and the loading of the nodes. (The derivation of the switched-capacitor equivalent resistance for low frequency inputs is covered in Appendix A, which results in the same $1/f_S C_S$ value.)

If we want to extend this modelling approach to multi-phase or non-reciprocal structures, a network of resistors is not helpful, as discussed earlier. Thus, a different perspective is necessary to address the issue. A unilateral approach is required to model the non-reciprocal behaviour.

To create a unilateral model, we can start by examining the average current flow at each node independently from the other nodes in the system. For example, for the two-phase switched-capacitor topology we have been working on (Fig. 2.1), rather than considering the average current flowing from V_1 to V_2 , let us first solely model the average current flowing into V_2 , which can be written as follows:

$$I_{A,2} = \frac{V_1 - V_2}{R} \tag{2.8}$$

where $I_{A,2}$ corresponds to the average current flowing into the node V_2 , and R has the same value as the one derived in Eqn. 2.7. Basically, we are focusing on the behaviour

from the end of ϕ_1 until the end of ϕ_2 . An ideal voltage buffer can be employed as shown in Fig. 2.5a to model $I_{A,2}$ without affecting the other nodes (i.e., V_1 node). The ideal voltage buffer does not load the node V_1 and realizes the desired unilateral behaviour. The loading of the node V_1 should also be accounted for to complete the model. The average current into the node V_1 can be written as follows using a similar approach:

$$I_{A,1} = \frac{V_2 - V_1}{R} \tag{2.9}$$

With this equation, we are focusing on the behaviour from the end of ϕ_2 until the end of ϕ_1 to complete the one sampling period. Thus, a second buffer + R branch can be placed, as shown in Fig. 2.5b. By combining two branches, the complete continuous-time model is obtained (Fig. 2.5c).

From this derivation, it follows that each phase change corresponds to a buffer + R branch, where C_S transfers voltage information from the previous node to the present node (modelled with the buffer), meanwhile charging itself to the present node voltage (modelled with R). Thus, the number of phases determines the number of buffer + R branches needed to complete the model. Once the model is complete, each node is connected to both a buffer and a resistor. (The derivation for low frequency inputs is covered in Appendix B, which results in the same buffer + R branch based model.)

The two buffer + R branches can be combined into a single R branch by using the Substitution Theorem, which leads back to the initial single equivalent resistance model (Fig. 2.1). This combination shows the compatibility of the new approach with the traditional one. This example also shows that it is possible to create a bilateral network by using multiple unilateral branches.

Although two buffer + R branches may seem redundant for a two-phase switchedcapacitor topology, the primary motivation of this work is to address topologies with non-reciprocal low frequency behaviour with multiple phases. Thus, this new approach provides us the tool that is needed to create unilateral continuous-time models.



Figure 2.5: Two-phase switched-capacitor topology a) with the average current into V_2 is modelled, b) with the average current into V_1 is modelled, and c) with its complete proposed continuous-time model.

2.1.3 Examples

2nd-Order Passive Switched-Capacitor Low-pass Filter with Real Poles

The 2^{nd} -order passive switched-capacitor filter shown in Fig. 2.6a has three phases, where C_S is connected to C_{I1} , C_{I2} , and ground nodes, respectively [3]. Three phases correspond to three buffer + R branches, which can be cascaded to create a loop as shown on the left side of Fig. 2.6b. Because C_S is connected to ground during ϕ_3 , further simplifications¹ can be performed on the model as shown on the right-hand side of Fig. 2.6b. The resulting continuous-time model clearly shows the low-pass filter operation with two real poles at $1/RC_{I1}$ and $1/RC_{I2}$ without the need for any further analysis.

Fig. 2.7 shows the simulated transfer function of the PSC filter discrete-time implementation and continuous-time model with $C_S = 200 fF$, $C_I = 8pF$, $f_S = 160 MHz$. PSC simulations are performed with SpectreRF periodic steady state and sampled periodic AC analyses [56]. Model simulations are performed with ac simulations. It can be seen that both transfer function curves closely follow each other until the Nyquist rate, i.e., $f_S/2$.

Fig. 2.8 shows the simulated step response of the PSC filter discrete-time implementation and the continuous-time model. The bottom plot shows the difference between the discrete-time implementation sampled response and the continuous-time model response. The error is dominant (with less than 5mV) during the initial transient phase, which corresponds to the high-frequency response with limited model accuracy. As the step response settles, i.e., for the DC response, the error becomes zero.

High-Order Passive Switched-Capacitor Low-Pass Filter with Real Poles

It is possible to increase the order of this filter simply by increasing the number of clock phases and C_I 's to which C_S is connected to before getting reset to ground. Fig. 2.9a

¹These further simplifications eliminate a buffer + R branch that is connected to the ground node. If the ground current of interest, the initial model can be kept as it is.



Figure 2.6: a) 2^{nd} -order PSC filter with three clock phases [3, p. 228], b) its continuous-time model.



Figure 2.7: 2^{nd} -order PSC filter simulated discrete-time implementation and continuoustime model frequency responses with logarithmic and linear scale frequency axes.



Figure 2.8: 2^{nd} -order PSC filter simulated discrete-time implementation and continuous-time model step responses, and the error between the step responses.



Figure 2.9: a) N^{th} -order PSC filter with N + 1 clock phases [4], b) its continuous-time model.

shows an N^{th} -order PSC filter schematic [4], and its continuous-time model is presented in Fig. 2.9b.

It should be noted that in this example, the model is applied to a topology where C_S is not only connected to the zero impedance voltage sources as ground but also to other capacitors (C_I 's). However, since C_I 's are integrating capacitors with values much larger than C_S [57] [4], they can be treated as low impedance nodes, and the model works with good accuracy. Chapter 4 provides further details on the applicability and limitations of the model.

2.2 The Continuous-Time Model for Non-Grounded Sampling Capacitors

The proposed modelling approach that is defined in Section 2.1 can address multi-phase switched-capacitor structures, which can also have non-reciprocal low frequency be-


Figure 2.10: a) Parasitic insensitive switched-capacitor integrator, and b) its incorrect continuous-time model.

haviour. However, it relies on grounded sampling capacitors. There are a variety of switched-capacitor circuit implementations that employ sampling capacitors with both plates switching in between different nodes. One typical example is the parasitic insensitive switched-capacitor integrator shown in Fig. 2.10a. Conventionally, the switching parts are modelled using negative resistance, -R, as shown in Fig. 2.10b [6, p. 417]. Although this model can capture the correct positive voltage gain value, -R indicates that for positive V_1 values, the current flows into the V_1 node from the virtual ground, which is not the case. Moreover, if this continuous-time approach is used to analyze the effect of the finite amplifier gain or the offset voltage, the analysis may result in false or inaccurate values together with a fictitious right-half plane pole at +1/[(1 + A)RC]rad/s, where A is the finite amplifier gain defined by $v_{out}/(v_+ - v_-)$. Apart from the parasitic insensitive switched-capacitor integrator, DC/DC converters with flying capacitors and even passive switched-capacitor filters can include sampling capacitors with non-grounded switching plates. This section generalizes the proposed continuous-time model for switched-capacitor circuits with non-grounded sampling capacitors to address the structures mentioned above.



Figure 2.11: a) Two-phase switched-capacitor topology with non-grounded sampling capacitor and b) its proposed continuous-time model.

2.2.1 Theory

A two-phase switched-capacitor topology where both plates are non-grounded is shown in Fig. 2.11a. The average current flows into the port V_2 in one clock period, I_{A2} , can be written using charge balance equations as:

$$I_{A2} = \frac{\Delta Q}{T_S} = \frac{C_S \cdot (V_1 - V_2)}{T_S} = (V_1 - V_2) f_S C_S$$
(2.10)

This current equation can be modelled using a voltage controlled voltage source (VCVS) whose value is equal to V_1 , connected in series with an equivalent resistance of R with the value of $1/f_SC_S$. Another branch of VCVS and R is necessary to model the average current flowing into the port V_1 to complete the model, as shown in Fig. 2.11b. This model can be easily adapted to multi-phase switched-capacitor structures by merely increasing the number of VCVS + R branches, similar to the approaches in Section 2.1. Then, the number of VCVS + R branches should be equal to the number of clock phases in the topology. A similar model for switched-capacitor structures with two phases was also previously investigated [58]. (The derivation of the model for voltage



Figure 2.12: Parasitic insensitive switched-capacitor integrator a) switching parts, b) its continuous-time model, and c) rearranged continuous-time model.

sources with input frequencies much smaller than the sampling frequency is similar to the proofs given in Appendices A and B.)

2.2.2 Examples

Parasitic Insensitive Switched-Capacitor Integrator

The generalized continuous-time model can be used to analyze the parasitic-insensitive switched-capacitor integrator with a non-grounded sampling capacitor. Fig. 2.12a shows the switching parts of the integrator, where V_1 corresponds to the input voltage, and V_2 corresponds to the virtual ground of the amplifier shown in Fig. 2.10a.

In Φ_1 , C_1 is connected between V_1 and ground nodes; this sampled voltage value,



Figure 2.13: The continuous-time model of the parasitic insensitive switched-capacitor integrator.

 $(V_1 - 0)$, becomes the value of the VCVS on the second branch (Fig. 2.10b). Whereas during Φ_2 , C_1 is connected between ground and V_2 nodes. Thus, the VCVS value in the first branch becomes $-V_2$. Since there exist two ground nodes, it is possible to rearrange the continuous-time model, as shown in Fig. 2.12c, where VCVSs can be shown by ideal inverting voltage buffers to have a more intuitive schematic. Fig. 2.13 shows the parasitic insensitive integrator, where the switching parts are replaced with the model.

It should be noted that this continuous-time SC integrator model can capture the correct sign of the voltage transfer function by the use of inverting voltage buffers rather than having a negative resistance. Furthermore, the model captures the correct input current direction. Thus, it can also be used to analyze the effects of finite amplifier gain and offset voltages without resulting in a fictitious right-half plane pole.

Flying Capacitor DC/DC Converter

The switched-capacitor DC/DC converter structure shown in Fig. 2.14a can be analyzed using the differential continuous-time model to determine the Thévenin equivalent of the switching parts. This structure involves a flying capacitor, C, and large capacitors, C_M 's for storing the DC value of the output voltage. The frequency of interest is much smaller than the sampling frequency [25, 26]. During ϕ_1 , C is connected between V_{DD} and V_{out} nodes, sampling a voltage of $V_{DD} - V_{out}$. Whereas, during ϕ_2 , C is connected between V_{out} and the ground node sampling a voltage of $V_{DD} - 0$ across itself. The top left schematic in Fig. 2.14b shows the differential switched-capacitor continuous-time model, where $R = 1/f_S C$. It is possible to further simplify the continuous-time structure by starting from the initial model and obtaining the Thévenin equivalent circuit shown in Fig. 2.14c. In order to find v_{out} from $2v_{out}$, the open circuit voltage, V_{DD} , and the output resistance, R/2, are divided by half. The conventional approach is to use a DC transformer model, as shown in Fig. 2.14d [26]. The conversion ratio, (m:n), and the output resistance, R_{OUT} , can be calculated to be the same as Fig. 2.14c. However, this approach is not intuitive, and the derivation is cumbersome.



Figure 2.14: a) Switched-capacitor DC/DC voltage converter, b) its simplified continuous-time model, c) the Thévenin equivalent of v_{out} obtained with the continuous-time model, d) conventionally used switched-capacitor converter transformer model.

Chapter 3

Complex-Conjugate Poles using Switches and Capacitors

The proposed continuous-time model offers an intuitive understanding of oversampled switched-capacitor topologies. The intuition provided by the continuous-time approach can also be used to develop new switched-capacitor topologies. In this section, complexconjugate poles are realized using passive switched-capacitor networks with the help of the continuous-time approach.

3.1 Passive Switched-Capacitor Butterworth Biquad

3.1.1 From Real Poles to Complex-Conjugate Poles

It is well known that passive RC networks can only form real poles [7]. Similarly, the creation of complex-conjugate poles in switched-capacitor circuits generally involves the use of active components to create a feedback loop able to move the real poles in the complex plane. In this section, it is shown that such a feedback loop can be created with a passive switched-capacitor network by exploiting time-variance and unilaterality highlighted by the proposed model.



Figure 3.1: a) Passive switched-capacitor filter with two real poles and b) its continuoustime model.

In Fig. 3.1a, a switched-capacitor low-pass filter with two real poles is shown. Although, in this case, two switches and the sampling capacitor, C_S , could be substituted by a simple resistor, the proposed continuous-time model is used to highlight the presence of a virtual loop (Fig. 3.1b).

The real poles of the circuit shown in Fig. 3.1b can be transformed into complexconjugate ones just by inverting the gain of the buffer in the return path (Fig. 3.2). This inversion creates a non-reciprocal network with a feedback path, which makes it possible to obtain complex-conjugate poles. These new poles have a natural oscillation frequency (ω_0) and a quality factor (Q) given by:

$$\omega_0 = \frac{\sqrt{2}}{R\sqrt{C_{I1}C_{I2}}}\tag{3.1}$$

$$Q = \frac{\sqrt{2C_{I1}C_{I2}}}{C_{I1} + C_{I2}} \le \frac{1}{\sqrt{2}}$$
(3.2)



Figure 3.2: Low-pass filter with complex-conjugate poles involving a negative feedback loop.

When Q becomes greater than 0.5 (assuming a C_{I1}/C_{I2} ratio in between 0.17 and 5.83), poles become complex-conjugate. For $C_{I1} = C_{I2}$, Q reaches its maximum value of $1/\sqrt{2}$ that corresponds to a 2^{nd} -order Butterworth filter transfer function. The reason for an upper bound in the achievable Q is due to the loop gain being limited to 1 in the given structure.

3.1.2 Implementation of the Passive Switched-Capacitor Filter

At this point, we have the continuous-time model of the filter with 2^{nd} -order Butterworth characteristics. How can we transfer this idea from the continuous-time model to an actual discrete-time filter implementation? One way to create the inverting behaviour is to use a differential structure, where the charge of the sampling capacitor, C_S , is crosstransferred between positive and negative terminals. This operation can be realized by the circuit shown in Fig. 3.3 based on four phases, two of which are labeled as ϕ_1 and ϕ_2 , and two of which are labeled negatively as ϕ_{1N} and ϕ_{2N} . During ϕ_1 and ϕ_2 phases, the circuit behaves like Fig. 3.1a, while during the negative phases, charge inversion is obtained. The output is sampled on both ϕ_2 and ϕ_{2N} . Thus, the circuit behaves as a two-phase system from the output perspective. This topology is similar to the one obtained in [44], where, however, four sampling capacitors are used. CHAPTER 3. COMPLEX-CONJUGATE POLES USING SWITCHES AND CAPACITORS 34



Figure 3.3: Differential 2^{nd} -order passive switched-capacitor filter with complex-conjugate poles.

Sanity Check with Four-Phase Continuous-Time Model

The filter shown in Fig. 3.3 has four phases and a cross-coupled structure. At first glance, it is not easy to verify the filter's operation and intuitively understand that the filter can give the desired characteristics dictated by the continuous-time model shown in Fig. 3.2. However, by applying the proposed model, the four-phase filter can be transformed into a continuous-time topology as a sanity check. A continuous-time model can be built by using buffer + resistor branches corresponding to each phase change $(\phi_1 \rightarrow \phi_2, \phi_2 \rightarrow \phi_{1N}, \phi_{1N} \rightarrow \phi_{2N}, \phi_{2N} \rightarrow \phi_1)$ as shown in Fig. 3.4a. The resistance value, R, is equal to $1/(f_S C_S)$, where f_S is defined for two phases, as the output is sampled at every two phases. This model can be simplified by merging parallel branches (Fig. 3.4b). Moreover, cross-coupled buffer branches can be replaced with inverter branches to obtain Fig. 3.4c, since the circuit operation is differential. Fig. 3.4c is the fully differential version of the circuit reported in Fig. 3.2. It should be noted here that if the common-mode analysis is required, cross-coupled branches should be left as they are (Fig. 3.4b).





Figure 3.4: Differential 2^{nd} -order Butterworth biquad continuous-time model: a) Direct representation of four phases, b) with parallel branches merged, c) cross-coupled buffers are replaced by inverter branches.

Conventional Discrete-Time Analysis

The pole locations of the differential 2^{nd} -order Butterworth passive switched-capacitor filter can be found by first deriving the filter transfer function in the z-domain and later obtaining the s-domain transfer function by using the bilinear approximation. The z-domain transfer function can be derived by writing the charge-domain equations as follows:

$$(C_{I1} + C_S)V_1\left[n - \frac{1}{3}\right] = C_{I1}V_1\left[n - \frac{4}{3}\right] - C_SV_{out} + Q_{in}[n]$$

$$(C_{I2} + C_S)V_{out}[n] = C_{I2}V_{out}[n - 1] + C_SV_1\left[n - \frac{1}{3}\right]$$

$$(3.3)$$
where $Q_{in}[n] = \int_{(n-1)T_S}^{nT_S} g_mV_{in}(t)dt$

where n represents the sampled value at the end of ϕ_2 . The resulting z-domain transfer function can be written as:

$$\frac{V_{out}(z)}{Q_{in}(z)} = \frac{z^2 \left[\frac{C_S}{(C_S + C_{I1})(C_S + C_{I2})}\right]}{z^2 - z \left[\frac{C_{I1}(C_S + C_{I2}) + C_{I2}(C_S + C_{I1}) - C_S^2}{(C_S + C_{I1})(C_S + C_{I2})}\right] + \frac{C_{I1}C_{I2}}{(C_S + C_{I1})(C_S + C_{I2})}$$
(3.4)

 ω_0 and Q of the filter can be determined by using the bilinear approximation, $z = \frac{1+sT_S/2}{1-sT_S/2}$, as follows:

$$\omega_0 = \frac{2f_S C_S}{\sqrt{C_{I1}C_S + C_{I2}C_S + 2C_{I1}C_{I2}}} \bigg|_{C_{I1,I2} \gg C_S/2} \approx \frac{\sqrt{2}}{R\sqrt{C_{I1}C_{I2}}}$$
(3.5)

$$Q = \frac{\sqrt{C_{I1}C_S + C_{I2}C_S + 2C_{I1}C_{I2}}}{C_{I1} + C_{I2} + C_S} \bigg|_{C_{I1,I2} \gg C_S/2} \approx \frac{\sqrt{2C_{I1}C_{I2}}}{C_{I1} + C_{I2}} \le \frac{1}{\sqrt{2}}$$
(3.6)

The calculated s-domain ω_0 and Q of the discrete-time filter become the same as the values derived by the continuous-time model in Eqn. 3.1 and 3.2 under the assumption of $C_{I1,I2} \gg C_S/2$.



Figure 3.5: Simulated magnitude responses for 1dB droop at 1MHz with $f_S = 160MHz$ of the two-real-pole and 2^{nd} -order Butterworth biquads.

3.1.3 Filter Transfer Function Simulations

The 2^{nd} -order Butterworth passive switched-capacitor biquad, the two-real-pole passive switched-capacitor biquad (Fig. 2.6), and their continuous-time models were simulated in 65 nm CMOS technology by using SpectreRF periodic steady state (pss) and periodic AC (pac) analyses. A sampling frequency of 160MHz was chosen for a filter passband of 1MHz. MOSFET transmission gates were used as switches with on resistances, R_{SW} , of $1k\Omega$ and off parasitic capacitance, C_{Prs} , of 1.8fF.

The simulated normalized magnitude transfer function with $C_{I1} = C_{I2} = 3.56pF$ and $C_S = 400fF$ for the real-pole filter and that with $C_{I1} = C_{I2} = 5.14pF$ and $C_S = 200fF$ for the Butterworth filter are reported in Fig. 3.5. The small discrepancy between the proposed model and the switched-capacitor implementation is due to the parasitic capacitances of the transmission gates, which shift the expected pole frequencies and also the DC gain ($\propto 1/(f_S C_S)$) of the filters. Note that the Butterworth filter provides more than 10dB suppression for frequencies above 7MHz compared to the real-pole filter. This additional suppression is thanks to the Butterworth's complex-conjugate poles resulting in sharper band edge filtering characteristics.

3.1.4 Effect of Parasitic Capacitances

The switches and capacitors (C_I and C_S) introduce parasitic capacitances C_{Prs} to the circuit. The top and bottom-plate parasitic capacitances of each capacitor are added on itself. C_{Prs} associated with switches, on the other hand, are added both on C_I and C_S .

For both of the filters, any C_{Prs} that was added only on C_I shifts the desired pole locations. For the Butterworth filter, as long as C_{I1} and C_{I2} values change similarly, the Q value is not affected by the added C_{Prs} (Eqn. 3.2), whereas the C_{Prs} of the switches added on C_S limits the minimum value of C_S . Choosing smaller size switches results in lower C_{Prs} . However, switch size is also determined by the required switch resistance R_{SW} , where R_{SW} is limited by the settling error. For a lower settling error, R_{SW} should be small, which requires larger switches. Thus, there is a trade-off between the settling error and minimum capacitance value. Moreover, the C_{Prs} of the switches changes with the signal level and introduces nonlinearity for large signal amplitudes.

The 2^{nd} -order Butterworth filter and the two-real-pole biquad were simulated using Spectre pss and pac with two tones at 10 and 19MHz in order to compare their nonlinearity performances. The intermodulation tone was placed close to the filter cut-off. Switch capacitances were the primary source of nonlinearity since ideal g_m -cells were used. The out-of-band 3^{rd} -order input intercept point (IIP3) of the 2^{nd} -order Butterworth was simulated at 29dBm, which compares favourably with the out-of-band IIP3 of the two-real-pole biquad that was simulated at 13dBm.

3.1.5 Single Pole Passive Switched-Capacitor Complex Filter

It is interesting to note that the proposed filter structure with complex-conjugate poles shown in Fig. 3.3 is very similar to the complex filter shown in Fig. 3.6 with quadrature inputs and a single sampling capacitor. Fig. 3.6a shows the complex passive switchedcapacitor filter reported by *Madadi et al.* that realizes a single complex pole by exploiting the in-phase (I), and quadrature (Q), i.e., 90° phase shifted input signals [8]. The current input of the filter is defined as $i_{in} = i_{in,I} + j \cdot i_{in,Q}$, whereas the voltage output of the filter is defined as $v_{out} = v_{out,I} + j \cdot v_{out,Q}$ [8]. The filter has four phases during which C_S rotates in between positive and negative I and Q nodes.

To derive the filter transfer function (v_{out}/i_{in}) , the continuous-time model can be developed as shown in Fig. 3.6b including four branches. Then, Kirchhoff's current law can be applied at the positive $V_{out,I}$ and $V_{out,Q}$ nodes:

$$i_{in,I} = \frac{v_{out,I}}{2} \cdot sC_I + \left(\frac{v_{out,I}}{2} - \frac{-v_{out,Q}}{2}\right) \frac{1}{R}, \quad i_{in,Q} = \frac{v_{out,Q}}{2} \cdot sC_I + \left(\frac{v_{out,Q}}{2} - \frac{v_{out,I}}{2}\right) \frac{1}{R} \quad (3.7)$$

where R is equal to T_S/C_S . By using Eqn. 3.7, it is possible to write the filter transfer function as follows:

$$(i_{in,I} + j.i_{in,Q}).2R = (v_{out,I} + jv_{out,Q})sRC_I + (v_{out,I} + jv_{out,Q}) + (v_{out,Q} - jv_{out,I})$$
(3.8)

$$TF(s) = \frac{v_{out,I} + jv_{out,Q}}{i_{in,I} + j.i_{in,Q}} = \frac{2R}{sRC_I + 1 - j}$$
(3.9)

From Eqn. 3.9, the single complex pole of the filter can be written as $(-1+j)/RC_I$ rad/s. Fig. 3.7 shows simulated normalized transfer functions of the discrete-time passive switched-capacitor filter and the continuous-time model with a sampling frequency of 160MHz, C_S of 200fF, and C_I of 8pF.



Figure 3.6: Single pole complex PSC filter a) discrete-time implementation [8] and b) its continuous-time model.



Figure 3.7: Simulated normalized frequency responses of PSC complex filter discrete-time implementation and its continuous-time model.

3.2 High-Order Filtering with Complex-Conjugate Poles

To achieve high-order low-pass filtering, designers traditionally cascade biquads having complex-conjugate pole pairs with different quality factors. However, for any filter order higher than 2, complex-conjugate poles with a quality factor greater than $1/\sqrt{2}$ are needed to generate for example Butterworth filter characteristics (the same is also valid for Chebyshev and Elliptic filters). However, this approach is not an option for the passive switched-capacitor biquad described in the previous section (Section 3.1) because the maximum achievable quality factor is limited to $1/\sqrt{2}$. One way to realize high-order filtering with larger than $1/\sqrt{2}$ quality factor is to use active devices, for example, g_m cells, in the feedback path [5]. However, this solution increases the noise and adds nonlinearity to the system along with the increased power consumption. Moreover, it may not be easy to scale the filter for newer technologies when we start to introduce active components. Thus, a new approach is needed to realize a sharp higher than 2^{nd} -order filtering characteristics implemented only with capacitors and switches.

3.2.1 Continuous-Time Design and Optimization

It is known that complex-conjugate poles can be generated by closing a feedback path around a cascade of stages with real poles. In Fig. 3.8, the feedback is realized by connecting the output of the cascade to the input through an inverting buffer and an equivalent resistance, R. This structure is also the N^{th} -order realization of the 2^{nd} -order Butterworth biquad filter reported previously in Section 3.1. Thanks to the continuoustime approach, it is easy to analyze how the number of branches affects the pole locations and changes the associated quality factors. To determine the pole locations, the transfer function of the filter shown in Fig. 3.8 can be written as follows:



Figure 3.8: The proposed high-order low-pass PSC filter with complex-conjugate poles.

$$\frac{v_{out}(s)}{i_{in}(s)} = \frac{R}{(1 + sRC_{I1})(1 + sRC_{I2})(1 + sRC_{I3})...(1 + sRC_{IN}) + 1}$$
(3.10)

Fig. 3.9 shows the locations of the poles generated by a cascade of 2, 3, and 4 elements with the same valued C_I 's for each stage. For a loop gain equal to one, which is the maximum achievable with a passive switched-capacitor network, the maximum quality factor is obtained starting from a cascade of real coincident poles, which demands identical C_I 's. The generated poles have the same quality factors as the poles of a Butterworth filter but with different frequencies (Fig. 3.9). The shift in the pole frequencies leads to in-band peaking in the filter transfer function that increases with the order of the filter. A recent independent work decreases the feedback coefficient -1 to reduce the in-band peaking. However, this solution results in a smoother filter transfer function close to the band edge [54]. Actually, from Fig. 3.8, it can be observed that if the feedback coefficient -1 is decreased until 0, the resulting filter becomes the same as the previously discussed all real pole filter (Fig. 2.9b).

3.2.2 Implementation of the High-Order Passive Switched-Capacitor Filter

A 3^{rd} -order filter is chosen to be designed as it can approximate the Butterworth filter behaviour with less than 1dB in-band peaking. Fig. 3.10 shows the continuous-time



Figure 3.9: Pole locations of the 2^{nd} , 3^{rd} , and 4^{th} -order filters, where the gray circles mark the same order Butterworth pole locations having the same 1dB cut-off frequency.

model of the 3^{rd} -order filter. An example of a 3^{rd} -order implementation is shown in Fig. 3.11. In this filter, C_S shares charge with C_{I1} , C_{I2} , and C_{I3} in ϕ_1 , ϕ_2 , and ϕ_3 , respectively, creating a unilateral signal flow. During the phase change from ϕ_3 to ϕ_1 , C_S is flipped to create the negative feedback. Note that Fig. 3.10 also shows the related phase changes corresponding to each branch.

The pole locations derived from the model can be verified by evaluating the z-domain transfer function of the filter shown in Fig. 3.11 as follows (assuming equal C_I 's for simplicity):

$$\frac{V_{out}(z)}{Q_{in}(z)} = \frac{1}{C_S} \frac{z^3 (1-\alpha)^3}{(z-\alpha)^3 + z^2 (1-\alpha)^3}$$
(3.11)

where α is $C_I/(C_I + C_S)$ and Q_{in} is the amount of charge fed into the filter in one sampling period, T_S . Although the network shown in Fig. 3.11 is periodic with $2T_S$, T_S is defined for three phases because the output is sampled in every three phases. The quality factor of the created poles can be found by mapping z-domain poles to s-domain using the bilinear approximation. As C_I/C_S ratio increases, Q of the complex-conjugate pair approaches to 1, which is the Q value predicted by the continuous-time model used for the synthesis of the filter.

Simulations of the proposed filter and the 3^{rd} -order Butterworth response (with the same DC gain and 1dB droop frequency) are compared in Fig. 3.12. Although the pole



Figure 3.10: The continuous-time model of the $3^{rd}\text{-}\mathrm{order}$ filter with complex-conjugate poles.



Figure 3.11: The PSC implementation of the 3^{rd} -order filter with complex-conjugate poles.



Figure 3.12: The transfer function of the PSC filter with complex-conjugate poles compared with the same order Butterworth and the 7^{th} -order real pole filter having the same 1dB droop frequency.

locations deviate from the ones of a Butterworth, the 3^{rd} -order filter maintains similar in-band flatness and selectivity. The figure inset shows that the response of the proposed filter deviates from the ideal Butterworth filter's response less than 1dB close to the filter band edge. Fig. 3.12 also compares the proposed solution with the 7^{th} -order all real pole passive switched-capacitor filter reported by Tohidian et al. [50] by assuming a maximum in-band drooping of 1dB. Even though the proposed filter has a lower order, a higher selectivity is achieved over a decade close to the filter pass-band thanks to the presence of complex-conjugate poles.

Chapter 4

Noise Analysis and Practical Limitations of the Model

The proposed continuous-time model inherits the assumptions of the traditional twophase switched-capacitor resistor equivalence: The driving node voltage should determine the voltage sampled by the sampling capacitor, C_s . As the sampling frequency goes to infinity, and the sampling capacitor value goes to zero, the equivalent resistor model works correctly. However, under practical conditions, limited settling time and charge sharing paths disturb this ideal behaviour. Moreover, as previously discussed in Chapter 2, the continuous-time model is only valid for input voltage sources with frequencies much smaller than the sampling frequency, i.e., for oversampled circuits. In this chapter, these limitations are studied to determine the accuracy of the proposed model when the conditions are only partially met.

4.1 Settling Time

A resistor in series with the sampling capacitor creates a non-zero time constant, which leads to settling error. This series resistance can be due to the switch on resistance or the source resistance. For a conventional switched-capacitor topology shown in Fig. 4.1,



Figure 4.1: Basic two-phase switched-capacitor topology with series resistances.

when switches have on resistances of R_{SW} , the equivalent resistance seen can be derived as follows:

$$R = \frac{V_1 - V_2}{I_A} = \frac{(V_1 - V_2)T_S}{\Delta Q} = \frac{(V_1 - V_2)T_S}{C\Delta V}$$
with $\Delta V = (V_1 - V_2)\frac{1 - e^{-\frac{1}{2f_S R_S W C_S}}}{1 + e^{-\frac{1}{2f_S R_S W C_S}}}$

$$R = \frac{1}{f_S C_S}\frac{1 + e^{-\frac{1}{2f_S R_S W C_S}}}{1 - e^{-\frac{1}{2f_S R_S W C_S}}}$$
(4.2)

where f_S is the sampling frequency and ΔV is the voltage difference on C_S between the end of ϕ_1 and the end of ϕ_2 phases. As R_{SW} goes to zero, R becomes $1/f_S C_S$, as expected. For example, if the settling time constant, $(R_{SW}C_S)$, is set to 15% of the sampling period, $1/f_S$, the error in the equivalent resistance becomes 7.40%, while if the time constant is set to 5% of the sampling period, the accuracy becomes 0.01%.

One way to relax this problem without requiring a higher clock frequency is to increase the sampling frequency of the switched-capacitor by introducing time-interleaved stages [46], as done in the presented prototype described in Chapter 5.

While the maximum clock frequency is limited due to the settling time of the switched capacitor circuit, the model's accuracy increases as the oversampling rate increases. Specifically, in the case where the clock frequency is fixed and satisfies the settling time requirement, then as the oversampling rate increases, the filter's bandwidth decreases and the accuracy of the model improves. However, a practical limitation here is that as



Figure 4.2: a) 1^{st} -order PSC filter and b) its continuous-time model with the traditional equivalent resistance approach [2].

the oversampling rate increases, the ratio of the sampling capacitors to the integrating capacitors will decrease and may lead to unreasonably small sampling capacitors.

4.2 Charge Sharing

In the case C_S is connected to a capacitive driving impedance, there is an issue of charge sharing between two capacitors that leads to an error in the equivalent resistor behaviour. For example, Fig. 4.2a shows the 1st-order passive switched-capacitor filter with current input and voltage output, where C_S is switching between an integrating capacitor, C_I , and a ground node. As mentioned in Chapter 1, C_I capacitor can be treated as a voltage source for C_I values much larger than C_S , and thus the switching parts can be replaced by an equivalent resistor with a value of $1/f_S C_S$ (Fig. 4.2b). However, when C_S becomes comparable to the driving impedance, C_I , the charge stored in C_S is not an accurate representation of the initial voltage on C_I . Moreover, the pole created (proportional to C_S/C_I) moves toward the higher frequencies, where the model accuracy is limited.

It is possible to investigate the continuous-time model accuracy for this 1st-order passive switched-capacitor filter. Fig. 4.3 shows the frequency transfer functions simulated using SpectreRF periodic steady state and periodic AC analyses for a sampling frequency



Figure 4.3: 1st-order PSC filter discrete-time implementation and continuous-time model simulated frequency responses for $C_I/C_S = 40$ and $C_I/C_S = 4$ with logarithmic and linear scale frequency axes.

of 160 MHz, $C_S = 200 fF$, and with C_I/C_S ratios of 40 and 4. For $C_I/C_S = 40$, the error in the 3dB cut-off frequency is 1%, whereas for $C_I/C_S = 4$ the error becomes 12%. (The notch at f_S is due to the inherent anti-aliasing property of the integration sampler [2].)

Fig. 4.5 shows the percentage error in the simulated 3dB cut-off frequency for changing C_S/C_I and f_{3dB}/f_S ratios. In Fig. 4.5a, the error approximately follows $C_S/2C_I$ for lower values of C_S/C_I , and as C_S becomes comparable to the driving impedance, C_I , the model accuracy decreases. For Fig. 4.5b, the error increases, as f_{3dB} becomes closer to the sampling frequency, f_S .

Fig. 4.4 shows the simulated step response of the PSC filter discrete-time implementation and the continuous-time model for C_I/C_S ratios of 40 and 4. The bottom plot shows the difference between the discrete-time implementation sampled and the



Figure 4.4: 1st-order PSC filter discrete-time implementation and continuous-time model simulated step responses for $C_I/C_S = 40$ and $C_I/C_S = 4$.

continuous-time model responses. The error is dominant during the initial transient phase, which corresponds to the high-frequency response with limited model accuracy. As the step response settles, i.e., for the DC response, error approaches to zero. As expected, the error is smaller for the C_I/C_S ratio equal to 40, when the model accuracy is better.

4.3 Accuracy of the Model in PSC Filter Response

The continuous-time model is valid for bandlimited sources with frequencies much smaller than the sampling frequency. However, it is shown that passive switched-capacitor filter frequency responses obtained using the continuous-time model closely follows the discretetime analysis up to Nyquist frequency, $f_S/2$, as long as the pole frequencies are much



Figure 4.5: Error in the 3dB cut-off frequency calculation of a 1st-order PSC filter using the continuous-time model for changing a) C_S/C_I and b) f_{3dB}/f_S ratios.

smaller than the sampling frequency [57, 59]. An example of this behaviour can also be seen in Fig. 4.3, where even at frequencies close to the sampling frequency of 160MHz, the -20dB/dec slope of the 1st-order filtering profile continues. This is because when the model accuracy starts to decrease at higher frequencies, the dominant impedance is determined by the large integrating capacitors present in the system. As $C_I \gg C_S$ and $f_{3dB} \ll f_S$, the model can predict the pass-band gain and the pole frequency with high accuracy. For the input frequencies closer to $f_S/2$, i.e., for frequencies higher than the pole frequency, C_I impedance starts to be the dominant impedance determining the overall frequency response. Thus, although at those frequencies the continuous-time model of the switching parts is not accurate, the continuous-time model of the filter results in the correct transfer function.

4.4 Noise Analysis and Simulation Using the Model

The proposed continuous-time model can also be used to analyze noise without the need for complex discrete-time charge balance analysis like the one used in [3]. The noise introduced by each buffer + resistor branch, i.e., in each phase change, is equal to the noise associated with the equivalent resistance R_{Eqv} (i.e., $4kTR_{Eqv}$). However, since, in each phase change, two switches are involved, $4kTR_{Eqv}$ noise must be divided into two uncorrelated noise sources with a power spectral density equal to $2kTR_{Eqv}$. Furthermore, the noise created by the same switch in two consecutive phase changes must be represented by two correlated noise sources with inverse polarity. This inversion can be explained by considering the noise injected during a clock phase when a switch is conducting. When the switch is on, it connects two capacitors, which store the injected noise with opposite polarities, as shown in Fig. 4.6.



Figure 4.6: Correlated noise charge stored in two capacitors with opposite polarities.



Figure 4.7: 2^{nd} -order PSC filter and its continuous-time model.

4.4.1 Examples

Passive Switched-Capacitor Low-pass Filter with Real Poles Analysis

As an example of the noise analysis, the 2^{nd} -order passive switched-capacitor filter with real poles is investigated. The filter's continuous-time model shown in Fig. 4.7 includes two resistors: one associated with the phase change from ϕ_1 to ϕ_2 and one associated with the phase change from ϕ_3 to ϕ_1 .

Fig. 4.8 shows noise modelling. Each resistor is associated with two uncorrelated $2kTR_{Eqv}$ noise sources, which are generated by the switches that C_S is connected to during the three related phases. Note that the noise associated with the ϕ_1 switch appears in both of the branches, and the two ϕ_1 noise sources are fully correlated with inverse polarity. From the circuit in Fig. 4.8, the output noise spectral density can be written as follows:



Figure 4.8: 2^{nd} -order PSC filter and its continuous-time model with noise sources, where the two ϕ_1 noise sources are fully correlated with inverse polarity.

$$V_{N,out}^{2} = 2kTR_{Eqv} \left[\left| \frac{1}{(1+j\omega R_{Eqv}C_{I1})(1+j\omega R_{Eqv}C_{I2})} - \frac{1}{1+j\omega R_{Eqv}C_{I2}} \right|^{2} + \frac{\phi_{3}}{(1+j\omega R_{Eqv}C_{I2})^{2}} + \frac{\phi_{3}}{(1+j\omega R_{Eqv}C_{I1})^{2}(1+j\omega R_{Eqv}C_{I2})^{2}} \right]$$
(4.3)

There are a couple of interesting observations that can be extracted from this output noise spectral density equation. First of all, ϕ_1 switch does not contribute to the in-band noise spectral density, where $f < f_{3dB}$. Only ϕ_2 and ϕ_3 switches contribute to the output noise at those frequencies to a total value of $4kTR_{Eqv}$. Moreover, this observation can be extended to the all real pole high-order passive switched-capacitor filter shown in Fig. 2.9. The in-band noise spectral density is always equal to $4kTR_{Eqv}$ with equal contributions from ϕ_N and ϕ_{N+1} switches, where Nis the filter order. Noise contribution from the other switches cancelled by inversely correlated noise sources in consecutive branches. This is also one of the advantages of the high-order charge-rotating passive switched-capacitor filter structure, i.e., as the filter order increases the noise contributions stays the same [4]. The second observation is that at higher frequencies, filter noise is dominated by the ϕ_2 and ϕ_1 switch noises, $2kTR_{Eqv}$ each, with a 1st-order decaying behaviour. Again, this observation can be extrapolated for the Nth-order filter, where dominant out-of-band noise



Figure 4.9: 2nd-order PSC filter discrete-time noise analyzes signal flow diagram.

sources become the N^{th} and $N - 1^{st}$ switches.

Fig. 4.9 shows the discrete-time noise analysis signal flow graph for the 2^{nd} -order passive switched-capacitor filter with real poles [3, p. 237] (where $\alpha_i = C_{Ii}/(C_{Ii} + C_S)$ and $\beta_i = 1 - \alpha_i$). It can be shown that the output noise spectral density obtained using the model is approximately equal to the value calculated using the discrete-time noise analysis when R_{Eqv} is replaced by $1/(f_S C_S)$.

Passive Switched-Capacitor Low-pass Filter with Real Poles Simulation

The continuous-time noise model can also be used for simulations. Especially for the design phase, the proposed noise modelling approach can shorten simulation times considerably.

Fig. 4.10 shows the schematic for the noise simulation. At first, noiseless resistors are used on the branches to generate two uncorrelated noise sources for a single resistor. Later, three separate noisy resistors are used to generate noise associated with each switch. Those noises are added to the model with the help of voltage controlled voltage sources. The polarity of the added noise sources only matters if the same noise source appears on more than one branches. In that case, it is important to reverse the polarity of the noise source on one of the consecutive branches. This inversion can be seen in between ϕ_1 noise sources in Fig. 4.10. The model is simulated using ac noise analysis for $C_I = 8pF$ and $R = 1/(f_S C_S)$ with a sampling frequency of 160MHz and $C_S = 200fF$. The simulation took 1.06 seconds. Fig. 4.11 shows the simulated output noise spectral density, which closely follows passive switched-capacitor filter noise simulations.

Passive switched-capacitor noise simulations are performed using SpectreRF periodic steady state (pss) and periodic noise (pnoise) analyses [56]. Switches are implemented as ideal switches with resistors in series. Switch resistances are chosen to be $1.3k\Omega$, such that clock phase over settling time constant (~ $R_{SW}.C_S$) ratio equals to 8, which leads to a low settling error of 0.03%. Although this simulation is mainly for verification purposes, choosing the switch resistance, R_{SW} , value is important. R_{SW} cannot be very large as that hurts the settling of the switched-capacitor. Also, for noise simulations, they cannot be too low. The reason is that to simulate the noise correctly for a sampled circuit, the folded noise should be accounted for correctly up to a couple of decades after the noise cut-off frequency ($1/2\pi R_{SW}C_S$). Thus, small R_{SW} values require long simulations times, or they may lead to incorrect results if simulation specifications entered poorly. With the given specifications, pss and pnoise simulations took around 12 minutes and 40 seconds. Compared to the previous 1.06 seconds, this simulation time is around 700 times longer (717 times longer to be exact), and with high-order filtering and time-interleaved structures, it can be even longer.

Table 4.1 shows the simulated spot noise contributions of each switch. 10 kHz and 6 MHz frequencies are chosen as values well below and well above the 3 dB cut-off frequency of the filter (410 kHz). Both ac noise and phoise simulation results are aligned with the expected dominant noise sources.



Figure 4.10: The continuous-time model of the 2^{nd} -order PSC filter for noise simulation.



Figure 4.11: Simulated output noise power spectral densities of 2^{nd} -order PSC filter and its continuous-time model.

	ac noise using CT model		pnoise using DT circuit	
	In-Band (10 kHz)	Out-of-Band (6 MHz)	In-Band (10 kHz)	Out-of-Band (6 MHz)
ϕ_1	0.01	49.44	0.01	49.09
ϕ_2	50.00	50.00	50.33	50.37
ϕ_3	49.99	0.56	49.66	0.54

Table 4.1: Percentage spot noise contributions of switches.



Figure 4.12: The 2^{nd} -order complex-conjugate-pole filter proposed model with noise sources, where noise sources associated with the same switch are fully correlated with inverse polarity.

Passive Switched-Capacitor Butterworth Filter Noise Analysis and Simulation

The noise analysis of the proposed 2^{nd} -order Butterworth biquad (Fig. 3.3) is performed by using the model in Fig. 4.12, which is developed as described in Section 4.4. Two of the four resistors presented in the model are associated with the phase change $\phi_1 \rightarrow \phi_2$ (for four-phase representation: $\phi_1 \rightarrow \phi_2$ or $\phi_{1N} \rightarrow \phi_{2N}$), and the other two resistors are associated with the phase change $\phi_2 \rightarrow \phi_1$ (for four-phase representation: $\phi_2 \rightarrow \phi_{1N}$ or $\phi_{2N} \rightarrow \phi_1$). Due to the differential nature of the filter, it is sufficient to investigate the noise associated with one of the ϕ_1 switches and one of the ϕ_2 switches. Later, the derived noise power can be multiplied by two to find the differential noise.

In Fig. 4.12, the noise sources associated with the ϕ_1 switch are placed in two consecutive branches that are connected to the V_1 positive terminal. Note that the two ϕ_1 noise sources are fully correlated with inverse polarity. Similarly, the noise sources associated with the ϕ_2 switch are placed in two consecutive branches that are connected to the V_{out} positive terminal. Note that the two ϕ_2 noise sources are fully correlated with inverse



Figure 4.13: Simulated output noise spectral densities of the two-real-pole and 2^{nd} -order Butterworth biquads for 1dB droop at 1MHz with $f_S = 160MHz$.

polarity. Using Fig. 4.12, the noise analysis is straightforward, and the output noise spectral density can be written as follows:

$$V_{N,out}^{2} = 2.2kTR_{Eqv} \left[\underbrace{\left[\frac{j\omega R_{Eqv}C_{I1}}{2 - \omega^{2}R_{Eqv}^{2}C_{I1}C_{I2} + j\omega R_{Eqv}(C_{I1} + C_{I2})} \right]^{2} + \underbrace{\left[\frac{\phi_{2}}{2 - \omega^{2}R_{Eqv}^{2}C_{I1}C_{I2} + j\omega R_{Eqv}(C_{I1} + C_{I2})} \right]^{2} \right]$$

$$(4.4)$$

The coefficient 2 is added to include the effect of the differential switches' noise. It can be shown that, when R_{Eqv} is replaced by $1/(f_S C_S)$, the output noise spectral density obtained using the continuous-time model is approximately equal to the value calculated using discrete-time noise analysis.

The simulated output noise spectral densities are shown in Fig. 4.13 with both the two-real-pole biquad and the 2^{nd} -order Butterworth discrete-time biquad sized the same as in the previous subsection. For both the real-pole and Butterworth biquads, simulated noise spectral densities in the passband are approximately equal to each other. Moreover, the simulation results closely follow the calculation results.



Figure 4.14: 2^{nd} -order PSC filter and its continuous-time model with noise sources, where two ϕ_1 noise sources are fully correlated with an inverse polarity.

Two-Phase Switched-Capacitor Structure

The proposed noise model can be applied to the conventional two-phase switched-capacitor structure. In this case, the model involves two uncorrelated noise sources, each having $2kTR_{Eqv}$ noise spectral density, as shown in Fig. 4.14. The resulting total $4kTR_{Eqv}$ noise spectral density value is different from the well-known 2kT/C integrated noise value of this structure. However, it is possible to reconcile these two values by calculating the noise spectral density over $0 - f_S/2$ bandwidth as below:

$$V_N^2 = \frac{2kT}{C} \frac{1}{f_S/2} = \frac{2kT}{C} \frac{2}{f_S} = \frac{4kT}{f_S C_S} \Big|_{\frac{1}{f_S C_S} = R_{Eqv}} = 4kT R_{Eqv}$$
(4.5)
Chapter 5

Prototype Filter Design and Implementation

This chapter describes the implementation of a passive switched-capacitor filter prototype with complex-conjugate poles, whose theory was previously laid out in Chapter 3. The 3^{rd} -order filter has a transfer function resembling that of a Butterworth filter. To the author's knowledge with this prototype, for the first-time, switched-capacitor complex-conjugate poles were integrated on silicon without the need of any amplifier and only using switches and capacitors. Moreover, the proposed continuous-time model in Chapter 2 is verified with the provided measurement results.

5.1 Prototype Filter Differential Implementation

Chapter 3 described how complex-conjugate poles can be generated using only switches and capacitors. An example 3^{rd} -order low-pass filter implementation was also shown in Fig. 3.11. Due to practical purposes, a differential implementation is preferred for the prototype filter. A fully differential 3^{rd} -order passive switched-capacitor filter was designed and fabricated in 0.13 µm TSMC CMOS process.

Fig. 5.1 shows the passive switched-capacitor filter implementation. The charge



Figure 5.1: The fabricated 3^{rd} -order PSC low-pass filter schematic and the timing diagram.

inversion of C_S is realized by exploiting the differential structure and cross-coupling of positive and negative nodes. The filter uses six phases: Three non-inverting phases (ϕ_1 , ϕ_2 , and ϕ_3), when top C_S is connected to the top C_I 's sequentially, while the bottom C_S is connected to the C_I 's on the bottom differential side. During the following three inverting phases (ϕ_{1N} , ϕ_{2N} , and ϕ_{3N}), C_S 's are connected to C_I 's, on the opposite sides.

In order to increase the sampling rate, three time-interleaved blocks, which consist of C_S 's and switches, are employed. Thus, the effective sampling period becomes the same as the period of a single phase.

The z-domain transfer function of the time-interleaved filter can be written as:

$$\frac{V_{out}(z)}{Q_{in}(z)} = \frac{2}{C_S} \frac{z(1-\alpha)^3}{(z-\alpha)^3 + (1-\alpha)^3}$$
(5.1)

where α is $C_I/(C_I + C_S)$, the coefficient 2 is due to the differential structure, and Q_{in} is the amount of charge that is fed into the filter in one sampling period. The z-domain transfer function reported in Eqn. 5.1 seems considerably different than the previously derived transfer function in Eqn. 3.11. The reason is that the sampling period, T_S , in Eqn. 5.1 is defined for a single phase due to the time-interleaved structure, whereas in Eqn. 3.11, T_S is defined for three phases. However, once mapped back to s-domain, it can be seen that Eqn. 3.11 and Eqn. 5.1 result in similar filtering characteristics for the same T_S .

The quality factor of the complex-conjugate pole pair can be calculated as 1 by mapping z-domain poles to s-domain using the bilinear approximation (assuming $C_I/C_S = 10$ and all C_I 's are equal to each other). As C_I/C_S ratio increases, Q of the complexconjugate pair approaches to 1. For a sampling frequency of 160MHz, the filter 3dBbandwidth was designed to be 470kHz. C_I 's were chosen to be equal for a maximum Q. All capacitors, C_I 's and C_S 's, were used as MIM capacitors with values 12.6pF and 225 fF, respectively, together with the added parasitic capacitances. Each sampling capacitor, C_S , is directly connected to six switches (Fig. 5.1), and 20% of the C_S 's are made of switch parasitic capacitances that are added on top of MIM capacitances. Although each integrating capacitor, C_I , is also directly connected to six switches, the effect of switch parasitic capacitances on C_I 's is negligible due to larger capacitance sizes. Smaller switches can lead to lower switch parasitic capacitances; however, there is a minimum limit on switch sizes due to the settling. The switches in this prototype were implemented using transmission gates and sized for $1.5k\Omega$ maximum on resistances for the operation range. A ring counter was implemented to produce six non-overlapping clock phases using an external clock signal at 160MHz. A g_m -cell was not included in the design in order not to dominate the noise and linearity responses of the passive switched-capacitor filter. Thus, the noise model can also be verified. In the prototype, the passive switched-capacitor filter is followed by an on-chip open drain output buffer to drive the probe used for the measurements. In order not to affect the noise measurement, the output buffer was designed to have lower noise spectral density compared to the filter.

5.1.1 Component Mismatch Effect on the Quality Factor

Component mismatch in between C_I 's or C_S 's can affect the filter transfer function by changing the filter cut-off frequency and the quality factor of the poles. It is possible to analyze these effects using the continuous-time model. As an example, a single-ended 2^{nd} order passive switched-capacitor filter can be investigated using the ω_0 and Q expressions previously derived in Section 3 as below (3.6):

$$\omega_0 = \frac{2f_S C_S}{\sqrt{C_{I1}C_S + C_{I2}C_S + 2C_{I1}C_{I2}}} \bigg|_{C_{I1,I2} \gg C_S/2} \approx \frac{\sqrt{2}}{R\sqrt{C_{I1}C_{I2}}}$$
(5.2)

$$Q = \frac{\sqrt{C_{I1}C_S + C_{I2}C_S + 2C_{I1}C_{I2}}}{C_{I1} + C_{I2} + C_S} \bigg|_{C_{I1,I2} \gg C_S/2} \approx \frac{\sqrt{2C_{I1}C_{I2}}}{C_{I1} + C_{I2}} \le \frac{1}{\sqrt{2}}$$
(5.3)

From the above equations, it can be calculated that for example for a 10% mismatch in between C_{I1} (+5%) and C_{I2} (-5%), Q decreases by 0.13%, which corresponds to a change from 0.707 to 0.706, i.e., the phase of the complex-conjugate poles changes from 45° to 44.93°. C_S mismatch in between the time-interleaved stages affects the R value. However, it does not affect the quality factor (see Eqn. 5.3).

5.2 Measurement Results

For transfer function measurements, a voltage source was fed into the filter through large external resistors (R_{Ext}) in series (a similar practice was reported in [60]). The voltage



Figure 5.2: The fabricated 3^{rd} -order PSC low-pass filter chip micrograph.

source together with R_{Ext} models the Thévenin equivalent of a g_m -cell with a finite output resistance (Fig. 5.1). The total external resistance is $200k\Omega$ (where the filter input resistance is $27.8k\Omega$, $R_{Eqv} = 1/f_SC_S$). The filter consumes $125\,\mu$ A from a 1.2Vpower supply, which is the power consumed by the phase clock generator. Fig. 5.2 shows the fabricated chip micrograph, where the active area of the chip is $0.06mm^2$ dominated by the integrating capacitors.

Fig. 5.3 shows the normalized transfer functions of the prototype measurements, post-layout schematic simulation, and the continuous-time model simulation. A good agreement is obtained between measurements, simulations, and the theory. For comparison, 7^{th} -order all real pole passive switched-capacitor filter simulation response is also added to the figure. The 7^{th} -order filter reported in [50] was designed to have the same DC gain and 1dB droop frequency. It can be observed that the 7^{th} -order filter transfer function has a much smoother roll-off around the cut-off frequency compared to other filter transfer functions, which causes more than 10dB attenuation loss around the band edge. Towards 10MHz, the measured filter response shows a flattening caused only by leakage on the PCB used for testing. External resistors lead to around 18dB attenuation on the PCB, which worsen the filtering profile by elevating the leakage level.



Figure 5.3: Transfer functions of the simulated and measured 3^{rd} -order low-pass PSC filter and the simulated 7^{th} -order all real pole filter.



Figure 5.4: Measured low-pass prototype filter transfer functions for changing sampling frequencies, f_S .

Fig. 5.4 shows the normalized measured transfer function of the filter for changing sampling frequency, f_S . It can be seen that the filter passband gain decreases for increasing f_S values, because the equivalent switched-capacitor resistance, $R(=1/f_SC_S)$, decreases, which is proportional to the low frequency gain of the filter as previously derived in Eqn. 3.10. Whereas, the filter cut-off frequency, which is proportional to the $1/RC_I(=f_SC_S/C_I)$, increases for increasing f_S values. Fig. 5.5 shows the first (v_{in}) , the second $(v_{2P} - v_{2N})$, and the third (v_{out}) order filtering nodes normalized measured transfer functions together with the continuous-time model simulation results, which match with very good agreement.

Output noise spectral density measurement results are shown in Fig. 5.6 with a



Figure 5.5: Measured low-pass prototype filter transfer functions for 1^{st} , 2^{nd} , and 3^{rd} -order output nodes for $f_S = 160 MHz$.

resolution bandwidth of 1Hz. The noise is measured at the output of the buffer and later referred to the output of the filter, v_{out} . At low frequencies, noise is equal to the noise of R_{Eqv} (i.e., $4kT/f_SC_S$). The solid black line shows the simulation result of the passive switched-capacitor filter, and the dashed line shows the continuous-time model simulation result. The noise spectral density was reported up to 1MHz due to the comparable noise floor of the spectrum analyzer. However, this measurement can show the in-band noise spectral density as well as the out-of-band decay close to the filter band edge. It can be seen that measured noise spectral density closely follows the simulations and the theory. The measured input referred noise is $15.8 \,\mu$ V, integrated between 10kHz and 470kHz. Out-of-band IIP3 is extrapolated using two blockers at 3.4MHz and 6.7MHz, creating an intermodulation product at 100kHz. Fig. 5.7 shows the input signal power and the 3^{rd} -order intermodulation distortion measurement results referred to the chip input (V_{in} in Fig. 5.1). The out-of-band IIP3 extrapolated is 55.1dBm leading to 92dB spurious-free dynamic range (SFDR).

Table 5.1 summarizes the measurement results and compares them with the 7^{th} order passive switched-capacitor filter [50] and also with the recently published 4^{th} -order
passive switched-capacitor filter that uses an active feedback structure [5]. The SFDR
obtained with this work is more than 20dB better with a much lower power consumption



Figure 5.6: Measured output noise spectral density over the prototype filter bandwidth.



Figure 5.7: Out-of-band IIP3 measurements of the prototype filter with two tones.

partially thanks to the presence of complex-conjugate poles, which allowed to reduce the filter order without compromising the filter selectivity close to the cut-off frequency. However, it should be noted that both other works employ a transconductance at the input, and [5] also uses another transconductance as the active feedback, which degrades filters' linearity and increases power consumption. Although the transconductance on the feedback path is not needed in this work thanks to the passive feedback topology, the input g_m -cell is only avoided due to testing purposes. Thus, Table 5.1 should be reviewed considering these differences. However, it can be possible to come up with an analytical comparison in between the reported 3^{rd} -order filter and the 7^{th} -order all real pole filter [50] when the g_m -cell is omitted, i.e., both filters are considered as current input - voltage output.

Analytical Comparison

The reported differential 3^{rd} -order filter with passive feedback and the differential 7^{th} order all real pole filter have the same input resistance, passband gain, in-band output

noise spectral density, if the sampling capacitor of the 3^{rd} -order filter is set half of the 7^{th} -order filter's sampling capacitor. This comparison is valid assuming the same sampling frequency for both of the filters with time-interleaving (Table 5.2).

Filters' noise and linearity performances should be considered to compare the SFDR. The same gain and output noise spectral density result in the same input referred noise. Moreover, since the current input filters both have the same input resistance and gain, they can be assumed to have similar linearity performance since the first-stage limits the linearity. Thus, the SFDR can be assumed to be the same for both of the filters.

As the final step, power consumption can be compared. The digital power consumption is proportional to the size and the number of switches and the sampling frequency assuming the same technology for both of the filters. The number of switches depends on the topology, and it is listed in Table 5.2 for differential implementations. The size of the switches can be assumed proportional to C_S , because C_I 's are much larger than C_S , i.e., the settling is determined by the C_S . Thus, digital power consumption is proportional to $C_S V_{DD}^2 N_{SW} f_S$, where N_{SW} is the number of switches. For our case, V_{DD} and f_S are the same for both of the filters. Thus, $C_S N_{SW}$ multiplication determines the power consumption, which results in around **86% power savings** in the waveform generator for the proposed 3^{rd} -order filter compared to the 7th-order all real pole implementation.

	This Work	[50]	[5]
Technology (nm)	130	65	180
Order	3	7	4
Poles	1 real, 2 compconj.	7 real	4 compconj.
Power (mW)	0.15	1.98	4.3
Wavegen. power (mW)	0.15	1.68	2.4
G_m -cells power (mW)	-	0.3	1.9
Voltage supply (V)	1.2	1.2	1.8
3dB cut-off (MHz)	0.47	0.4-30	0.49 - 13.3
Sampling rate (MS/s)	160	800	65-300
OB IIP3 (dBm)	55.1	11.7	15.03
Int. noise (μV)	15.8^{*}	13.7^{**}	13.6^{***}
IRN $(nV/\sqrt{H}z)$	23.3	4.57	6.54
SFDR (dB)	92'	64	68
Active area (mm^2)	0.06	0.42	2.9

Table 5.1: Summary results and comparison.

Integrated over *10k-470kHz, **50k-9MHz, ***100k-4.4MHz 'SFDR=2/3.(IIP3-Int. Noise)

Table 5.2: Analytical comparison of 3^{rd} -order filter with passive feedback and 7^{th} -order all real pole filter.

	This Work	[50]
Sample frequency	f_S	f_S
Sampling capacitor	C/2	C
Diff. input resistance	$2/f_S C$	$2/f_S C$
Gain (v_{out}/i_{in})	$2/f_S C$	$2/f_S C$
Output noise (V^2/Hz)	$8kT/f_SC_S$	$8kT/f_SC_S$
# of interleaved stages	3	8
Total $\#$ of switches	36	128

Chapter 6

Conclusions

This thesis reports a simplified model for oversampled passive switched-capacitor filters where the sampling parts of the circuit are replaced with continuous-time equivalents consisting of resistors and ideal buffers. The main advantage of the model is that it is composed of unilateral branches, and thus, it can address structures with non-reciprocal low frequency behaviour. Moreover, it can also be applied to structures with sampling capacitors with both terminals switching. The reported continuous-time model results in easy and intuitive analysis compared to the conventional charge-balance equations and leads to transfer function and noise analysis with good accuracy. Furthermore, the model can also be used for transfer function and noise simulations, which can require much shorter simulation times compared to conventional discrete-time simulations. An example noise simulation comparison in SpectreRF (pss+pnoise) and Spectre results in a decrease in the simulation time from 12 minutes 40 seconds to 1.06 seconds (around 700 times improvement). This thesis also covers some of the limitations and applicability of the model.

Through the use of this model, a passive switched-capacitor filter with complexconjugate poles is realized, which results in a sharper filter profile compared to its realpole counterparts. A 3^{rd} -order filter prototype with complex-conjugate poles was designed and fabricated in $0.13 \,\mu\text{m}$ CMOS process. Measurements performed on the filter prototype result in state-of-the-art performance meanwhile verifying the continuous-time modelling and design approach. Noise measurements and model simulations matched with less than 1dB accuracy. With this prototype, for the first time, switched-capacitor complex conjugate poles have been integrated on silicon without the need of any active circuitry. This represents a remarkable result that makes it possible to obtain sharp filtering profiles using passive switched-capacitor filters while showing that the design approach can lead to the invention of novel structures.

6.1 Contributions

The contributions of this thesis can be summarized as below:

- A new continuous-time modelling approach for oversampled switched-capacitor circuits was introduced [59]. The model can be used for analysis, design, and simulations.
- The continuous-time modelling approach was generalized for topologies with nongrounded sampling capacitors [61].
- Limitations of the model were studied [61].
- A 2nd-order passive switched-capacitor filter with passive feedback was introduced to realize complex-conjugate poles [59]. (Although, later we realized that there was a similar approach in the literature back in 1980 [44].)
- A 3rd-order passive switched-capacitor filter prototype with complex-conjugate poles was designed, fabricated, and tested [57].
- The continuous-time modelling approach was verified with signal and noise measurements on the fabricated prototype [57].



Figure 6.1: a) A symbolic passive switched-capacitor filter with a gain of 2 charge pump, and b) its continuous-time model.

6.2 On-Going and Future Work

6.2.1 Increasing Configurability and Linearity of the PSC Filter

There is on-going work together with Da Kang and QingNan Yu to increase the configurability and linearity of the passive switched-capacitor filter. As a capstone project, Da and QingNan (and also Yueqi Chen) designed a prototype passive switched-capacitor filter. The filter is configurable for 2^{nd} and 3^{rd} -order by only changing the clocking scheme and with the use of an external digital control bit. Moreover, the filter has gain and attenuation by two control, which is realized by making use of charge-pump structures within the sampling capacitor. Fig. 6.1a shows a symbolic passive switched-capacitor filter with the charge pump structure, where two sampling capacitors switch in between parallel and series configurations to provide a passive voltage gain of two from node v_1 to v_out . Fig. 6.1b shows the associated continuous-time model.

In order to increase the linearity of the filter, we focused on the input transconduc-



Figure 6.2: A voltage input voltage output passive switched-capacitor band-pass filter continuous-time model and its discrete-time implementation.

tance. Since the passive switched-capacitor filter is highly linear, the input transconductor becomes the linearity bottleneck. To address this issue with the contributions from Javid Musayev, we quantized the input g_m -cell and also the configurable passive switched-capacitor filter itself similar to the approach reported in [62].

The chip was fabricated in $0.18\,\mu\mathrm{m}$ CMOS process, and at this point measurements are on-going.

6.2.2 Band-Pass and High-Pass Passive Switched-Capacitor Filters

It is possible to obtain real pole band-pass and high-pass filters starting from the continuoustime model and then designing the passive switched-capacitor structures. An example band-pass filter with voltage input voltage output is shown in Fig. 6.2. If the output is defined as the current through C_{I2} , the structure shown in Fig. 6.2 becomes a voltage input current output 2^{nd} -order high-pass filter. These filters may find applications in biosensing, especially for applications where large undesired DC signals accompany small desired AC biosignals at the IC input. However, it should be noted that since these are voltage input filters, they do not have inherent anti-aliasing property, which is different from the current input passive switched-capacitor filters (see Fig. 1.4).

6.2.3 Passive Switched-Capacitor Digital-to-Analog Converter

It is possible to use the passive switched-capacitor filter as a digital-to-analog converter with reconstruction filtering. The simple idea is shown in Fig. 6.3, which is comparable to the RF-DAC receiver reported in [21], but possibly may result in better performance due to the high-order passive switched-capacitor filtering. The number of gray boxes including C_S and switches depends on the number of bits in the DAC. In the first step, C_S 's are either connected to the ground node or V_{DD} node depending on the digital control signals. In ϕ_2 , all C_S 's are connected to C_I 1. Since all C_S 's are also connected to each other at this phase, the charge sharing in between C_S 's leads to the digital to analog conversion. Moreover, C_S 's are connected to C_{I1} led to 1st-order filtering of the resulting analog signal. By increasing the number of phases and C_I 's, the filter order can be increased. In Fig. 6.3, the topology has 3rd-order characteristics with all real poles. Konstantinos Vasilakopoulos has been expanding this idea for an RF-DAC in his Ph.D. thesis.

6.2.4 High-Order Butterworth Passive Switched-Capacitor Filter

To realize complex-conjugate poles, in this thesis, we have added a single global feedback path to the passive switched-capacitor structure. As a result, although we can get complex-conjugate poles with large quality factors for high filter orders (Fig. 3.8 and Fig.3.9), the in-band peaking limits the use of those high-order filters. However, using a single-feedback can be just the beginning. It may be possible to obtain Butterworth filter characteristics for higher than 2^{nd} -order filters by using multiple feedback and feedforward paths and by adding charge-pump structures. Fig. 6.4 shows an example continuous-



Figure 6.3: The basic idea for a passive switched-capacitor DAC with 3^{rd} -order reconstruction filtering.

time model drawing for a 3^{rd} -order filter with multiple feedback loops. Multiple feedback and feedforward loops are implementable using multiple C_S 's. Also, the analysis can be possible using the reported continuous-time model rather than going through rigorous charge-domain analyses. Moreover, it may even be possible to use supervised learning to reach to a transfer function that can approximate the Butterworth filter behaviour. The author is planning to continue working on this idea as a hobby project.



Figure 6.4: Passive switched-capacitor filter continuous-time model with multiple feed-back loops.

Appendix A

Equivalent Resistance Derivation for Low Frequency

In Section 2.1 Eqn. 2.5 - 2.7 derive the equivalent switched-capacitor resistance by assuming ideal DC voltage sources connected to V_1 and V_2 nodes and by calculating the average current transfer in one clock period. While this derivation is quite well known for DC signals, to the author's knowledge, there is no equivalent derivation for low frequency ac signals. In this appendix, the equivalent impedance is derived for ac voltage sources with an input signal frequency much lower than the sampling frequency, f_s .

Assume that V_1 and V_2 nodes in Fig. 2.1 are connected to two voltage sources with maximum frequencies well below the sampling frequency, f_S , and switches are ideal with zero on resistances. Then, the current supplied by V_1 can be written as follows:

$$i_1(t) = i_{1pulse}(t) + i_{1cap}(t)$$
 (A.1)

where $i_{1pulse}(t)$ is the current pulse occurs at the instant the switch is closed at ϕ_1 , and $i_{1cap}(t)$ is the current supplied to the capacitor during ϕ_1 , as the voltage V_1 slowly varies. These two components can be written as below:

$$i_{1pulse}(t) = C_S[v_1(t) - v_2(t)] \sum_{n=-\infty}^{\infty} \delta(t - nT_S)$$

$$i_{1cap}(t) = C_S \frac{dv_1(t)}{dt} \sum_{n=-\infty}^{\infty} a_n e^{jn\omega_S t}$$
(A.2)
with $a_n = \begin{cases} \frac{2}{n\pi} (-1)^{\frac{n-1}{2}} & \text{n is odd} \\ 0 & \text{n is even and } n \neq 0 \\ 0.5 & n=0 \end{cases}$

where $i_{1pulse}(t)$ is a series of current pulses whose area is equal to $C_S.\Delta V (= C_S v_1(nT_S) - C_S v_2(nT_S))$, and $i_{1cap}(t)$ is the capacitor current equation multiplied by the rectangular pulse train, which is represented by its Fourier series. Note that the rectangular pulse train has the same phase of ϕ_1 with an amplitude of 1.

To define an impedance to model the topology as a linear time-invariant (LTI) system, the resulting current in response to the V_1 and V_2 voltages should be examined. From equations A.1 and A.2, it is possible to write the Fourier transform of the produced current as:

$$I_{1}(\omega) = \frac{C_{S}}{T_{S}} \sum_{k=-\infty}^{\infty} \left[V_{1}(j\omega - jk2\pi f_{S}) - V_{2}(j\omega - jk2\pi f_{S}) \right]$$

$$+ j\omega C_{S} \sum_{k=-\infty}^{\infty} a_{k}V_{1}(j\omega - jk2\pi f_{S})$$
(A.3)

where a_k are the Fourier series coefficients and equal to a_n for n = k. It can be observed from Eqn. A.3 that the current produced has a fundamental harmonic and also higher harmonics at the multiples of f_S . To model the circuit behaviour for low frequencies, only the fundamental harmonic would be of interest [63], which can be written as below:

$$I_{1Fund}(\omega) = f_S C_S [V_1(j\omega) - V_2(j\omega)] + j\omega \frac{C_S}{2} V_1(j\omega)$$
(A.4)

It can be seen that I_{1Fund} results in $1/f_S C_S$ resistance in between V_1 and V_2 nodes

and $C_S/2$ capacitance in between the V_1 node and ground. Although $C_S/2$ is needed for the exact representation at the fundamental frequency, for simplicity, it is ignored in the body of this thesis, as all the cases that are covered, sampling capacitors are connected to low impedance nodes, where the effect of $C_S/2$ becomes negligible. Moreover, it should be noted that $2/j\omega C_S$ impedance is much greater than $1/f_S C_S$ for the frequencies of interest $(\pi f \ll f_S)$. For a small set of cases where C_S is comparable to C_I , including $C_S/2$ might help to get results with better accuracy, which is briefly covered in Appendix C.

The impedance seen from the V_2 node can be derived similarly by writing the current equation sunk into the V_2 node, $i_2(t)$. In this case, the Fourier Transforms of the current produced can be written as follows:

$$I_{2}(\omega) = e^{-\frac{j\omega}{2f_{S}}} \frac{C_{S}}{T_{S}} \sum_{k=-\infty}^{\infty} \left[V_{2}(j\omega - jk2\pi f_{S}) - V_{1}(j\omega - jk2\pi f_{S}) \right]$$

$$+ j\omega C_{S} \sum_{k=-\infty}^{\infty} b_{k} V_{2}(j\omega - jk2\pi f_{S})$$
(A.5)
with $b_{k} = \begin{cases} \frac{2}{n\pi} (-1)^{\frac{n+1}{2}} & \text{n is odd} \\ 0 & \text{n is even and } n \neq 0 \\ 0.5 & n=0 \end{cases}$

whose fundamental component becomes:

$$I_{2Fund}(\omega) = e^{-\frac{j\omega}{2f_S}} f_S C_S[V_2(j\omega) - V_1(j\omega)] + j\omega \frac{C_S}{2} V_2(j\omega)$$
(A.6)

The exponential term in I_{2Fund} expression is due to the sampling phase shift, and it can be ignored for the frequencies of interest ($\pi f \ll f_S$). Once the exponential term is ignored, it is seen that I_{2Fund} expression results in a $1/f_SC_S$ resistance in between V_1 and V_2 nodes and a $C_S/2$ valued capacitance in between V_2 node and ground. This capacitance can be ignored as well because the impedance of $2/j\omega C_S$ is much greater than $1/f_SC_S$ for the frequencies of interest ($\pi f \ll f_S$). Thus, the voltage and current relationship between V_1 and V_2 voltage sources can be modelled by an equivalent resistance of $1/f_S C_S$ in between those two nodes. (As mentioned earlier, for a small set of cases where C_S is comparable to C_I , including $C_S/2$ might help to get results with better accuracy, which is briefly covered in Appendix C.)

Appendix B

Continuous-Time Model Derivation for Multi-Phase Topology

In Section 2.1, buffer + R model is derived by assuming ideal DC voltage sources connected to V_1 and V_2 nodes and by calculating the average current transfer in one clock period (Eqn. 2.9). In this appendix, the proposed multi-phase switched-capacitor continuous-time model is derived for ac voltage sources with an input signal frequency much lower than the sampling frequency, f_s .

Assume that V_1 , V_2 , and V_3 in Fig. B.1a are all connected to ideal voltage sources with input frequencies much smaller than the sampling frequency, f_S , and switches are ideal with zero on resistances. Then, the current supplied by V_1 can be written as follows:

$$i_{1}(t) = i_{1pulse}(t) + i_{1cap}(t)$$

$$i_{1pulse}(t) = C_{S}[v_{1}(t) - v_{3}(t)] \sum_{n=-\infty}^{\infty} \delta(t - nT_{S})$$

$$i_{1cap}(t) = C_{S} \frac{dv_{1}(t)}{dt} \sum_{n=-\infty}^{\infty} c_{n} e^{jn\omega_{S}t}$$

$$with c_{n} = \begin{cases} -\frac{2}{n\pi} \sin(\frac{n\pi}{3}) & n \in \mathbb{Z} \\ \frac{1}{3} & n=0 \end{cases}$$
(B.1)



Figure B.1: A three-phase switched-capacitor topology, and b) its continuous-time model.

It can be shown that these current equations lead to a resistor between V_1 and V_3 nodes with a value of $1/f_S C_S$ for the frequencies of interest. Moreover, Eqn. B.1 also leads to a $C_S/3$ valued capacitance in between the V_1 node and ground (the derivation and assumptions are similar to the ones discussed previously in A).

To complete the model and to replace the switching parts, all current components $(i_1, i_2, \text{ and } i_3)$ should be considered. It can be shown that, for the frequencies of interest, there is a general current equation valid for i_1 , i_2 , and i_3 that can be written as follows in the Fourier domain:

$$I_k(\omega) = f_S C[V_k(j\omega) - V_{Prev}(j\omega)]$$
(B.2)

where I_k represents the Fourier transform of the current component sourced by V_k voltage source (where k = 1, 2, and 3), and V_{Prev} represents the voltage source that C_S is connected to in the prior phase before connecting to V_k . It is important to note that although there exist three voltage sources, the current component related to each node only depends on two voltage sources that C_S is connected (one is at the relevant phase, and the other one is at the phase prior to the relevant phase).

Fig. B.1b shows the schematic representation of Eqn. B.2 employing ideal voltage controlled voltage sources (VCVS) in series with the $1/f_SC_S$ valued equivalent resistances. This schematic can be modified to have a more intuitive view, where VCVS's are replaced with ideal voltage buffers.

Appendix C

The Continuous-Time Model with Capacitance

As discussed in Appendices A and B, adding capacitance to the continuous-time model may lead to more accurate results, especially for the cases when C_S values are comparable to C_I . In this appendix, the two-phase 1st-order passive switched-capacitor filter is investigated with and without the added capacitance to the model. Fig. C.1 shows the filter's discrete-time implementation together with the two continuous-time models. Due to the two-phase structure of the filter, $C_S/2$ valued capacitance is added in between node V_{out} and ground (see Appendix A).

Fig. C.2 shows the frequency transfer functions simulated using SpectreRF periodic steady state (pss) and periodic AC (pac) analyses for a sampling frequency of 160 MHz, $C_S = 200 fF$, and with C_I/C_S ratios of 40 and 4. For the equivalent resistance model shown in Fig. C.1b with $C_I/C_S = 40$, the error in the 3dB cut-off frequency is 1%, whereas for the model with $C_S/2$ (Fig. C.1c), this error becomes almost 0%. For the equivalent resistance model with $C_I/C_S = 4$, the error in the 3dB cut-off frequency becomes 12%, whereas for the model with $C_S/2$, this error becomes 0.4%.

Fig. C.3 shows simulated step responses of the PSC filter discrete-time implementa-



Figure C.1: a) 1^{st} -order PSC filter, b) its equivalent resistance continuous-time model, and c) its equivalent resistance continuous-time model with $C_S/2$ capacitance added.



Figure C.2: 1st-order PSC filter simulated frequency responses of discrete-time implementation and continuous-time models with and without $C_S/2$ for $C_I/C_S = 40$ and $C_I/C_S = 4$ (logarithmic scale frequency axis on top and linear scale frequency axis on bottom).

tion and continuous-time models with C_I/C_S ratios of 40 and 4. The mid and bottom plots show the difference between the discrete-time implementation sampled and the continuous-time model responses. The error is dominant during the initial transient phase, which corresponds to the high frequency response with limited model accuracy. Also, it should be noted that the model with $C_S/2$ results in a much smaller error compared to the equivalent resistor model without $C_S/2$.



Figure C.3: 1st-order PSC filter simulated step responses of discrete-time implementation and continuous-time models with and without $C_S/2$ for $C_I/C_S = 40$ and $C_I/C_S = 4$.

Bibliography

- James Clerk Maxwell. Comparison of the electrostatic with the electromagnetic units. In A Treatise on Electricity and Magnetism, volume 2, chapter 19, pages 384–386. Oxford: The Clarendon Press, 2 edition, 1881.
- [2] A. Mirzaei, S. Chehrazi, R. Bagheri, and A. A. Abidi. Analysis of first-order antialiasing integration sampler. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 55(10):2994–3005, Nov 2008.
- [3] A. Winoto and B. Nikolic. Discrete time processing of rf signals. In Multi-Mode/Multi-Band RF Transceivers for Wireless Communication. New York, NY, USA: Wiley, 2011.
- [4] M. Tohidian, I. Madadi, and R. B. Staszewski. A 2mw 800ms/s 7th-order discretetime iir filter with 400khz-to-30mhz bw and 100db stop-band rejection in 65nm cmos. In 2013 IEEE International Solid-State Circuits Conference Digest of Technical Papers, pages 174–175, Feb 2013.
- [5] P. Payandehnia, H. Maghami, H. Mirzaie, M. Kareppagoudr, S. Dey, M. Tohidian, and G. C. Temes. A 0.49–13.3 mhz tunable fourth-order lpf with complex poles achieving 28.7 dbm oip3. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 65(8):2353–2364, Aug 2018.
- [6] David A. Johns and Ken Martin. Analog Integrated Circuit Design. John Wiley and Sons, Inc., 1 edition, 1997.

- [7] G. C. Temes and J. W. LaPatra. In Introduction to Circuit Synthesis and Design, page 255–256. New York, NY, USA: McGraw-Hill, 1977.
- [8] I. Madadi, M. Tohidian, and R. B. Staszewski. Analysis and design of i/q chargesharing band-pass-filter for superheterodyne receivers. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 62(8):2114–2121, Aug 2015.
- [9] G. C. Temes. Yannis tsividis, path-breaking researcher and educator. *IEEE Solid-State Circuits Magazine*, 6(4):48–51, Fall 2014.
- [10] A. Sheikholeslami. Capacitor as a resistor [circuit intuitions]. IEEE Solid-State Circuits Magazine, 9(3):7–59, Summer 2017.
- [11] J. T. Caves, S. D. Rosenbaum, M. A. Copeland, and C. F. Rahim. Sampled analog filtering using switched capacitors as resistor equivalents. *IEEE Journal of Solid-State Circuits*, 12(6):592–599, Dec 1977.
- [12] B. J. Hosticka, R. W. Brodersen, and P. R. Gray. Mos sampled data recursive filters using switched capacitor integrators. *IEEE Journal of Solid-State Circuits*, 12(6):600–608, Dec 1977.
- [13] R. W. Brodersen, P. R. Gray, and D. A. Hodges. Mos switched-capacitor filters. Proceedings of the IEEE, 67(1):61–75, Jan 1979.
- [14] K. Martin. Improved circuits for the realization of switched-capacitor filters. IEEE Transactions on Circuits and Systems, 27(4):237–244, April 1980.
- [15] T. C. Choi, R. T. Kaneshiro, R. W. Brodersen, P. R. Gray, W. B. Jett, and M. Wilcox. High-frequency cmos switched-capacitor filters for communications application. *IEEE Journal of Solid-State Circuits*, 18(6):652–664, Dec 1983.
- [16] H. M. Sandler and A. S. Sedra. Programmable switched-capacitor low-pass ladder filters. *IEEE Journal of Solid-State Circuits*, 21(6):1109–1119, Dec 1986.

- [17] J. Khoury and M. Banu. Yannis tsividis' early contributions to mos filters. *IEEE Solid-State Circuits Magazine*, 6(4):36–40, Fall 2014.
- [18] C. W. Solomon. Switched-capacitor filters: precise, compact, inexpensive. IEEE Spectrum, 25(6):28–32, June 1988.
- [19] C. Liu, S. Chang, G. Huang, and Y. Lin. A 10-bit 50-ms/s sar adc with a monotonic capacitor switching procedure. *IEEE Journal of Solid-State Circuits*, 45(4):731–740, April 2010.
- [20] and P. R. Gray and B. Nikolic. A 14-b 12-ms/s cmos pipeline adc with over 100-db sfdr. *IEEE Journal of Solid-State Circuits*, 39(12):2139–2151, Dec 2004.
- [21] P. E. Paro Filho, M. Ingels, P. Wambacq, and J. Craninckx. An incremental-chargebased digital transmitter with built-in filtering. *IEEE Journal of Solid-State Circuits*, 50(12):3065–3076, Dec 2015.
- [22] B. P. Ginsburg and A. P. Chandrakasan. 500-ms/s 5-bit adc in 65-nm cmos with split capacitor array dac. *IEEE Journal of Solid-State Circuits*, 42(4):739–747, April 2007.
- [23] C. Andrews and A. C. Molnar. A passive mixer-first receiver with digitally controlled and widely tunable rf interface. *IEEE Journal of Solid-State Circuits*, 45(12):2696– 2708, Dec 2010.
- [24] R. Bagheri, A. Mirzaei, S. Chehrazi, M. E. Heidari, M. Lee, M. Mikhemar, W. Tang, and A. A. Abidi. An 800 mhz-6 ghz software-defined wireless receiver in 90-nm cmos. *IEEE Journal of Solid-State Circuits*, 41(12):2860–2876, Dec 2006.
- [25] M. D. Seeman and S. R. Sanders. Analysis and optimization of switched-capacitor dc_dc converters. *IEEE Transactions on Power Electronics*, 23(2):841–851, March 2008.

- [26] M. S. Makowski and D. Maksimovic. Performance limits of switched-capacitor dc_dc converters. In *Power Electronics Specialists Conference*, 1995. PESC '95 Record., 26th Annual IEEE, volume 2, pages 1215–1221 vol.2, June 1995.
- [27] J. B. Hughes, N. C. Bird, and R. S. Soin. Self-tuned rc-active filters for vlsi. *Electronics Letters*, 22(19):993–994, Sep. 1986.
- [28] Y. Tsividis, M. Banu, and J. Khoury. Continuous-time mosfet-c filters in vlsi. IEEE Journal of Solid-State Circuits, 21(1):15–30, Feb 1986.
- [29] H. Khorramabadi and P. R. Gray. High-frequency cmos continuous-time filters. *IEEE Journal of Solid-State Circuits*, 19(6):939–948, Dec 1984.
- [30] S. Pavan, Y. P. Tsividis, and K. Nagaraj. Widely programmable high-frequency continuous-time filters in digital cmos technology. *IEEE Journal of Solid-State Circuits*, 35(4):503–511, April 2000.
- [31] A. Selvakumar and A. Liscidini. Current-recycling complex filter for bluetooth-lowenergy applications. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 62(4):332–336, April 2015.
- [32] A. C. W. Wong, M. Dawkins, G. Devita, N. Kasparidis, A. Katsiamis, O. King, F. Lauria, J. Schiff, and A. J. Burdett. A 1 v 5 ma multimode ieee 802.15.6/bluetooth low-energy wban transceiver for biotelemetry applications. *IEEE Journal of Solid-State Circuits*, 48(1):186–198, Jan 2013.
- [33] K. Reddy, S. Rao, R. Inti, B. Young, A. Elshazly, M. Talegaonkar, and P. K. Hanumolu. A 16-mw 78-db sndr 10-mhz bw ct δσ adc using residue-cancelling vco-based quantizer. *IEEE Journal of Solid-State Circuits*, 47(12):2916–2927, Dec 2012.
- [34] H. Shibata, R. Schreier, W. Yang, A. Shaikh, D. Paterson, T. C. Caldwell, D. Alldred, and P. W. Lai. A dc-to-1 ghz tunable rfδσadc achieving dr=74 db and

bw=150 mhz at f_0 =450 mhz using 550 mw. *IEEE Journal of Solid-State Circuits*, 47(12):2888-2897, Dec 2012.

- [35] I. Fabiano, M. Sosio, A. Liscidini, and R. Castello. Saw-less analog front-end receivers for tdd and fdd. *IEEE Journal of Solid-State Circuits*, 48(12):3067–3079, Dec 2013.
- [36] J. A. Fredenburg and M. P. Flynn. A 90-ms/s 11-mhz-bandwidth 62-db sndr noiseshaping sar adc. *IEEE Journal of Solid-State Circuits*, 47(12):2898–2904, Dec 2012.
- [37] K. Abdelhalim, H. M. Jafari, L. Kokarovtseva, J. L. P. Velazquez, and R. Genov. 64channel uwb wireless neural vector analyzer soc with a closed-loop phase synchronytriggered neurostimulator. *IEEE Journal of Solid-State Circuits*, 48(10):2494–2510, Oct 2013.
- [38] H. Gao, R. M. Walker, P. Nuyujukian, K. A. A. Makinwa, K. V. Shenoy, B. Murmann, and T. H. Meng. Hermese: A 96-channel full data rate direct neural interface in 0.13μm cmos. *IEEE Journal of Solid-State Circuits*, 47(4):1043–1055, April 2012.
- [39] P. Lajevardi, A. P. Chandrakasan, and H. Lee. Zero-crossing detector based reconfigurable analog system. *IEEE Journal of Solid-State Circuits*, 46(11):2478–2487, Nov 2011.
- [40] Z. Zhu and W. Bai. A 0.5-v 1.3-μWanalog front-end cmos circuit. IEEE Transactions on Circuits and Systems II: Express Briefs, 63(6):523–527, June 2016.
- [41] G. Martinelli and M. Salerno. Commutated capacitor bank as a filter resonator. *Electronics Letters*, 14(14):427–428, July 1978.
- [42] G. Martinelli and M. Salerno. Passive approach to switched-capacitor circuits. Circuit Theory and Applications, 8:325–335, July 1980.

- [43] S. Manetti and A. Liberatore. Realization of non-reciprocal networks using switchedcapacitors. In *IEEE International Symposium on Circuits and Systems*, pages 969– 972, April 1980.
- [44] S. Manetti and A. Liberatore. Switched-capacitor lowpass filter without active components. *Electronics Letters*, 16(23):883–885, November 1980.
- [45] A. Liberatore, S. Manetti, and A. Ricci. On the synthesis of passive switched capacitor low pass filters. In *IEEE International Symposium on Circuits and Systems*, pages 737–740, May 1982.
- [46] M. Ghaderi, J. Nossek, and G. Temes. Narrow-band switched-capacitor bandpass filters. *IEEE Transactions on Circuits and Systems*, 29(8):557–572, August 1982.
- [47] S. Manetti. Passive switched capacitor filters: general biquad topology. *Electronics Letters*, 20(2):101–102, January 1984.
- [48] F. J. Wang, G. C. Temes, and S. Law. A quasi-passive cmos pipeline d/a converter. *IEEE Journal of Solid-State Circuits*, 24(6):1752–1755, Dec 1989.
- [49] Iman Madadi. Fully Integrated SAW-Less Discrete-Time Superheterodyne Receiver.PhD thesis, Delft University of Technology, 10 2015. An optional note.
- [50] M. Tohidian, I. Madadi, and R. B. Staszewski. Analysis and design of a highorder discrete-time passive iir low-pass filter. *IEEE Journal of Solid-State Circuits*, 49(11):2575–2587, Nov 2014.
- [51] M. Tohidian, I. Madadi, and R. B. Staszewski. 3.8 a fully integrated highly reconfigurable discrete-time superheterodyne receiver. In 2014 IEEE International Solid-State Circuits Conference Digest of Technical Papers (ISSCC), pages 1–3, Feb 2014.

- [52] Y. Xu and P. R. Kinget. A switched-capacitor rf front end with embedded programmable high order filtering and a +15dbm ob-b1db. In 2015 IEEE Radio Frequency Integrated Circuits Symposium (RFIC), pages 291–294, May 2015.
- [53] Y. Xu and P. R. Kinget. A switched-capacitor rf front end with embedded programmable high-order filtering. *IEEE Journal of Solid-State Circuits*, 51(5):1154– 1167, May 2016.
- [54] P. Payandehnia, H. Maghami, M. Kareppagoudr, and G. C. Temes. Passive switchedcapacitor filter with complex poles for high-speed applications. *Electronics Letters*, 52(19):1592–1594, 2016.
- [55] J. Yli-Kaakinen, V. Lehtinen, and M. Renfors. Multirate charge-domain filter design for rf-sampling multi-standard receiver. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 62(2):590–599, Feb 2015.
- [56] Ken Kundert. Simulating switched-capacitor filters with spectrerf. 01 2006.
- [57] S. Z. Lulec, D. A. Johns, and A. Liscidini. A 150-μw 3rd-order butterworth passiveswitched-capacitor filter with 92 db sfdr. In 2017 Symposium on VLSI Circuits, pages C142–C143, June 2017.
- [58] M. Keskin, N. Keskin, and G. C. Temes. An efficient and accurate dc analysis technique for switched-capacitor circuits. *Analog Integrated Circuits and Signal Processing*, 30(3):239–241, Mar. 2002.
- [59] S. Z. Lulec, D. A. Johns, and A. Liscidini. A simplified model for passive-switchedcapacitor filters with complex poles. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 63(6):513–517, June 2016.

- [60] A. Pirola, A. Liscidini, and R. Castello. Current-mode, wcdma channel filter with in-band noise shaping. *IEEE Journal of Solid-State Circuits*, 45(9):1770–1780, Sep. 2010.
- [61] S. Z. Lüleç, D. A. Johns, and A. Liscidini. A third-order integrated passive switchedcapacitor filter obtained with a continuous-time design approach. *IEEE Transactions* on Circuits and Systems I: Regular Papers, pages 1–10, 2019.
- [62] J. Musayev and A. Liscidini. Quantized analog rx front-end for saw-less applications. In ESSCIRC 2018 - IEEE 44th European Solid State Circuits Conference (ESSCIRC), pages 306–309, Sep. 2018.
- [63] A. Mirzaei, H. Darabi, J. C. Leete, and Y. Chang. Analysis and optimization of direct-conversion receivers with 25duty-cycle current-driven passive mixers. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 57(9):2353–2366, Sep. 2010.